

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Heger, Dörte

### Working Paper Work and Well-Being of Informal Caregivers in Europe

Ruhr Economic Papers, No. 512

**Provided in Cooperation with:** RWI – Leibniz-Institut für Wirtschaftsforschung, Essen

*Suggested Citation:* Heger, Dörte (2014) : Work and Well-Being of Informal Caregivers in Europe, Ruhr Economic Papers, No. 512, ISBN 978-3-86788-587-4, Rheinisch-Westfälisches Institut für Wirtschaftsforschung (RWI), Essen, https://doi.org/10.4419/86788587

This Version is available at: https://hdl.handle.net/10419/103327

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



## WWW.ECONSTOR.EU



# RUHR ECONOMIC PAPERS



#### Imprint

#### Ruhr Economic Papers

Published by

Ruhr-Universität Bochum (RUB), Department of Economics Universitätsstr. 150, 44801 Bochum, Germany

Technische Universität Dortmund, Department of Economic and Social Sciences Vogelpothsweg 87, 44227 Dortmund, Germany

Universität Duisburg-Essen, Department of Economics Universitätsstr. 12, 45117 Essen, Germany

Rheinisch-Westfälisches Institut für Wirtschaftsforschung (RWI) Hohenzollernstr. 1-3, 45128 Essen, Germany

#### Editors

Prof. Dr. Thomas K. Bauer
RUB, Department of Economics, Empirical Economics
Phone: +49 (0) 234/3 22 83 41, e-mail: thomas.bauer@rub.de
Prof. Dr. Wolfgang Leininger
Technische Universität Dortmund, Department of Economic and Social Sciences
Economics - Microeconomics
Phone: +49 (0) 231/7 55-3297, e-mail: W.Leininger@wiso.uni-dortmund.de
Prof. Dr. Volker Clausen
University of Duisburg-Essen, Department of Economics
International Economics
Phone: +49 (0) 201/1 83-3655, e-mail: vclausen@vwl.uni-due.de
Prof. Dr. Roland Döhrn, Prof. Dr. Manuel Frondel, Prof. Dr. Jochen Kluve
RWI, Phone: +49 (0) 201/81 49-213, e-mail: presse@rwi-essen.de

#### Editorial Office

Sabine Weiler

RWI, Phone: +49 (0) 201/81 49-213, e-mail: sabine.weiler@rwi-essen.de

#### Ruhr Economic Papers #512

Responsible Editor: Jochen Kluve

All rights reserved. Bochum, Dortmund, Duisburg, Essen, Germany, 2014

ISSN 1864-4872 (online) - ISBN 978-3-86788-587-4

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editors.

Ruhr Economic Papers #512

Dörte Heger

Work and Well-Being of Informal Caregivers in Europe



## Bibliografische Informationen der Deutschen Nationalbibliothek

Die Deutsche Bibliothek verzeichnet diese Publikation in der deutschen Nationalbibliografie; detaillierte bibliografische Daten sind im Internet über: http://dnb.d-nb.de abrufbar.

http://dx.doi.org/10.4419/86788587 ISSN 1864-4872 (online) ISBN 978-3-86788-587-4

#### Dörte Heger<sup>1</sup>

## Work and Well-Being of Informal Caregivers in Europe

#### Abstract

Informal caregivers provide valuable services to elderly persons with long-term care needs, but the consequences of caregiving on caregivers are not yet fully understood. This paper illustrates the interrelation between caregiving and caregivers' labour force participation, cognitive ability, and health in a simple theoretical model, and estimates the effects of caregiving using panel data from thirteen European countries, which allows to analyze the effect of institutions on caregivers' outcomes. The results show that caregiving severely and signicantly reduces caregivers' probability of being employed, but only in countries with few formal care alternatives. Furthermore, caregivers in all countries suffer from worse mental health when caregiving is prompted by poor parental health. The results for the effects of caregiving on physical health and cognitive ability are mixed.

JEL Classification: I12, J14, J18, J22

Keywords: Informal care; labour supply; cognitive ability; physical and mental health

October 2014

<sup>1</sup> RWI. – I am extremely grateful to Steven Lehrer, Vincent Pohl, and Tammy Schirle for their valuable comments. This paper uses data from SHARE wave 4 release 1.11, as of March 28th 2013 (DOI: 10.6103/SHARE.w4.111) or SHARE wave 1 and 2 release 2.6.0, as of November 29 2013 (DOI: 10.6103/SHARE.w1.260 and 10.6103/SHARE data collection has been primarily funded by the European Commission through the 5th Framework Programme (project QLK6-CT:2001-00360 in the thematic programme Quality of Life), through the 5th Framework Programme (project QLK6-CT:2001-00360 in the thematic programme Quality of Life), through the 5th Framework Programme (project QLK6-CT:2001-00360 in the thematic programme Quality of Life), through the 5th Framework Programme (project QLK6-CT:2001-00360 in the thematic programme Quality of Life), through the 5th Framework Programme (project QLK6-CT:2001-00360 in the thematic programme Quality of Life), through the 5th Framework Programme (project QLK6-CT:2001-00360 in the thematic programme Quality of Life), through the 5th Framework Programme (projects SHARE-13, RII-CT-2006-062193, COMPARE, CIT5-CT-2005-028857, and SHARE-1EAP, N° 227822 and SHARE M4, N° 2619821. Additional funding from the U.S. National Institute on Aging (UID AG09740-1352, P01 AG005842, P01 AG08291, P30 AG12815, R21 AG025169, Y1-AG-4553-01, IAG BSR06-11 and OGHA 04-064) and the German Ministry of Education and Research as well as from various national sources is gratefully acknowledged (see www.share-project.org for a full list of funding institutions)- All correspondence to: Dörte Heger, RWI, Hohenzollernstr. 1-3, 45128 Essen, Germany, e-mail: doerte.heger@rwi-essen.de

#### 1 Introduction

When a parent's health declines, adult children often volunteer their time to assist the parent with personal care or household chores. Such informal care far exceeds formal care in the number of hours of care provided, as well as in terms of monetary value (OECD, 2005). While the need for informal care is growing quickly due to the increase in life expectancy, the supply of informal caregivers is decreasing due to low birthrates, because children tend to live further away from their parents, and because labour market participation among women, who traditionally account for the vast majority of informal caregivers, is increasing (Colombo et al., 2011). In response to the growing need for care, policymakers in several countries have aimed to encourage informal caregiving to reduce the financial pressure on public long term care (LTC) systems. However, the effects of caregiving on caregivers are not yet fully understood.

To better understand these effects, I analyze the impact of care provision to an elderly parent on adult daughters aged 50 to 70 in three important areas: labour force participation, cognitive ability, and health. These outcomes represent valuable indicators of caregivers' quality of life, well-being, and risk of financial vulnerability and are of relevance both to the individual caregivers and to society. While mature caregivers make up a significant portion of informal caregivers in the European Union (Glendinning et al., 2009), they have a lower attachment to the labour force and may be more prone to negatively react to the stress of caregiving than younger caregivers. In addition, parental caregiver than care provision to a friend or neighbour due to close emotional ties between parents and their children.<sup>2</sup>

This paper proceeds in two steps. First, I present the trade off between leisure time, caregiving activities, and work in a simple theoretical framework and relate the daughter's decision to provide care to individual characteristics, the parent's need for care, and the availability of formal LTC options. Second, I estimate the causal effects of caregiving on the caregiver while carefully controlling for the endogeneity of caregiving using longitudinal data from the Survey of Health, Ageing, and Retirement (SHARE) collected in 2004/2005, 2006/2007, and 2011/2012. This data set provides rich demographic and socio-economic information of the population 50+ in Europe and offers the unique opportunity to study the influence of institutional characteristics.

As caregiving may depend on unobserved individual characteristics, such as the emotional closeness to the care receiver or the caregiver's sense of duty, controlling for the endogeneity of the caregiving decision is paramount. I use a fixed effects instrument variable (FE-IV) approach to account for time-constant as well as time-varying endogeneity, which provides a more careful treatment of the endogeneity of caregiving than previous cross-sectional studies allow for. Furthermore, I analyze the influence of the intensity

<sup>&</sup>lt;sup>2</sup>Caregiving to parents-in-law is not considered due to data limitations.

of caregiving on caregivers' outcomes as measured by the frequency of care and by the institutional caregiving context. Specifically, I separately consider daily, weekly, and any frequency of caregiving and test for heterogeneous effects of caregiving in "formal care countries" with relatively generous formal care alternatives and "family care countries" where elderly care is predominantly provided by the family. Comparing the effects of caregiving in family and formal care countries will increase our understanding of the influence of the institutional background on the effects of caregiving on informal caregivers and inform the policy debate about the optimal provision of LTC. Moreover, to the best of my knowledge, this study is the first to look at cognitive ability of parental caregivers in a longitudinal, multi-country context.

To preview my results, I find caregiving decreases employment rates in family care countries by 34 to 60 percentage points depending on the frequency of care but has no impact on caregivers' employment probability in formal care countries. Contrary, caregiving increases the number of depressive symptoms in both family and formal care countries by nearly 3 and 4 additional symptoms, respectively. The effects on physical health and cognitive ability are mixed and vary with the institutional background.

The remainder of this paper is organized as follows. Section 2, summarizes the possible pathways through which caregiving may affect caregivers and links the caregiving context to the intensity of caregiving. Section 3 introduces the theoretical framework, Section 4 discusses the estimation strategy, Section 5 describes the SHARE data, and Section 6 presents the results. Section 7 concludes.

#### 2 Background and literature

Given the often considerable time commitment of informal caregiving and the potential loss of productivity if caregivers reduce their labour supply, the effect of caregiving on labour market outcomes has received much attention. Labour market outcomes for mature caregivers are especially important since caregivers close to the retirement age are less attached to the labour force than younger caregivers. As early retirement is generally linked to a reduction in pension benefits, understanding whether caregiving and work are compatible for elderly caregivers is important to evaluate caregivers' risk of financial vulnerability.

Theoretically, the direction of the effect of caregiving on labour market outcomes is uncertain. Caregiving may reduce employment or hours worked if caregivers substitute work time with care time, or if increased absenteeism of caregivers to fulfill caregiving obligations leads to job loss (Heitmueller and Inglis, 2007). Yet, labour market participation may increase if work offers respite from caregiving or if caregivers require additional income (Carmichael and Charles, 1998, 2003). Empirically, determining causal effects, however, is challenging due to reverse causality. Persons with less attachment to the labour market or a low time cost are more likely to become caregivers (Ettner, 1995, 1996; Heitmueller, 2007; Carmichael et al., 2010). In addition, labour force participation and caregiving may be influenced by the same unobserved characteristics, which would produce a spurious correlation. For example, individuals with high (unobserved) ability may be more productive employees and may also be more productive at providing informal care (He and McHenry, 2013). As a result, cross-sectional studies may not always be able to detect causal relationships. In a review of the literature, Lilly et al. (2007) conclude that caregiving is generally associated with a moderate reduction in hours worked, intensive caregiving leads to negative consequences for both hours worked and labour force participation. Infrequent caregivers, however, are sometimes found to be less likely to retire (Carmichael and Charles, 1998; Dentinger and Clarkberg, 2002). While most of the analyzed studies rely on cross-sectional data and are thus subject to above mentioned concerns, the results have been largely mirrored by recent longitudinal studies (Heitmueller, 2007; King and Pickard, 2013; van Houtven et al., 2013).

However, the vast majority of evidence relies on U.S. or U.K. data, two countries with very "liberal" welfare regimes (Haberkern and Szydlik, 2010).<sup>3</sup> For continental Europe, the effects of caregiving may not necessarily be the same since these countries have different welfare arrangements and the effect of caregiving on labour supply under different institutional settings is still ambiguous. Ciani (2012) finds small or insignificant negative effects on employment of co-residential caregiving, with a slightly larger effect in Southern European countries in some specifications, while Viitanen (2010) finds caregiving only significantly reduces women's employment probability in Germany.<sup>4</sup> Contrary, Crespo and Mira (2010) find a negligible effect of daily parental caregiving on employment in Northern and Central European countries but a 50% decline in the employment probability in Southern European countries. Looking at cross-sectional data, Bolin et al. (2008b) find a negative effect on hours worked in Central European countries, but no effect on employment, whereas Crespo (2006) finds caregiving reduces women's probability of employment by 30-40% with slightly more adverse effects in Central Europe than in Southern Europe.<sup>5</sup> My paper complements and extends these studies by addressing time-varying and time-constant sources of endogeneity, analyzing different frequencies of care and institutional settings, studying a longer time span, and including two Eastern European countries, Poland and the Czech Republic.

Similar to labour market outcomes, interest in the health effects of caregiving is not limited to the individual caregiver. While informal caregiving helps to reduce public expenditures in the short run, long run effects are uncertain as negative health outcomes for caregivers may increase their health care needs in the future. Moreover, poor health may inhibit caregivers to tend to their caregiving duties.

 $<sup>^{3}</sup>$ Notable exceptions that control for the endogeneity of caregiving are Casado-Marín et al. (2011) and Meng (2013) who study co-residential caregiving in Spain and Germany, respectively.

<sup>&</sup>lt;sup>4</sup>For single women, Viitanen (2010) also finds significant effects in Greece, Italy, and the Netherlands.

<sup>&</sup>lt;sup>5</sup>Bolin et al. (2008b) also find a positive effect of caregiving on wages for women; men experience negative effects on employment and, in Central European countries, negative effects on hours worked.

Whether to expect positive or negative effects of caregiving on caregivers' physical and mental health is again ambiguous in theory. For one, according to the stress process model proposed by Aneshensel (1995), caregiving may lead to a decline in health due to increased stress or a change in health behaviour or diet in response to a reduction in personal time. Physiological responses to stress consistent with this theory have been documented in both medical and psychological studies (see, for example, Vitaliano et al. (2007)). For another, caregiving may increase physical activity and lead to health improvements. This view is expressed by the "healthy caregiver hypothesis" (Fredman et al., 2008). Further, caregiving can be rewarding as it conveys the feeling that the caregiver is needed, yet it might lead to loneliness if caregiving prevents other social activities, which can lead to positive or negative effects for mental health, respectively. Which effect dominates is still debated (see Pinquart and Sörensen (2003, 2007) for an overview). While Coe and van Houtven (2009) find continued caregiving decreases self-rated health and increases depressive symptoms over time, Brown et al. (2009) find decreased mortality rates of caregivers compared to non-caregivers. Moreover, health effects are influenced by the intensity of caregiving and caregivers' characteristics, with intensive caregivers and caregivers from lower socioeconomic background generally experiencing worse outcomes (Schulz et al., 1997; Hirst, 2005; Pinquart and Sörensen, 2005).

Cognitive effects of caregiving have so far received little attention in studies using large representative population surveys. However, like health, cognitive ability is of significant personal and public interest since preserving cognitive ability is essential for a self-determined and independent life and should thus be considered as an outcome of interest. Higher levels of stress have been shown to impact cognitive ability negatively, but caregiving may also provide cognitive stimulation and hence have positive effects on cognitive ability (Vitaliano et al., 2007; Lee et al., 2004; Bertrand et al., 2012). The empirical evidence is still inconclusive and more research is needed.<sup>6</sup> Positive results on cognitive ability are generally found for social participation and volunteering (Glei et al., 2005; Green et al., 2008; Engelhardt et al., 2010; Dobrescu and Christelis, 2012). Though caregiving is sometimes included as one possible volunteering activity, caregiving differs substantially from participating in a political, cultural or religious organization or club (Hsu, 2007). In particular, caregiving is usually motivated by the care receivers need for care and social norms and is thus likely to be more stressful than other volunteering activities. Hence, the effects of caregiving on cognitive ability may differ from the effects reported for social engagement.

#### 2.1 Institutions and caregiver burden

While the frequency of care provides an intuitive measure of the care intensity and more frequent caregiving has been associated with worse outcomes for caregivers as outlined above, institutional dif-

<sup>&</sup>lt;sup>6</sup>Ambiguous results are found in studies focusing on broadly defined spousal caregiving, while mostly negative effects are found for caregivers to persons suffering from dementia; see, for example, Caswell et al. (2003), de Vugt et al. (2006), Leipold et al. (2008), and Mackenzie et al. (2009).

ferences have been less studied due to data limitations but may be equally important in explaining differential effects of caregiving. For one, Matire and Schulz (2012) emphasize that the magnitude and direction of the effects of caregiving are influenced by caregivers' appraisal of the caregiving demands and adaptive capacities. When caregivers feel capable of dealing with the caregiving demands, positive outcomes on cognitive ability and health are more likely than when caregivers feel overwhelmed. Thus, the institutional setting may influence caregivers' appraisal of caregiving demands, for example, by determining the extent to which formal care options are available or by offering financial support for informal caregivers. For another, formal care alternatives may reduce negative consequences for caregivers by directly reducing the caregiving burden (Bass et al., 1996).

To analyze the influence of the intensity of caregiving, I separate countries into family and formal care countries based on their spending level on LTC. This grouping corresponds to separating countries into what are commonly thought of as countries with strong formal LTC, and countries with strong family based LTC. Haberkern and Szydlik (2010) evaluate countries based on whether and to what extent there is a legal obligation to support relatives in need, the services that the state provides or funds, and public opinion about whether the state or the family should in general be responsible for the care and support of dependent elderly people. Countries with strong formal-care systems include the Scandinavian countries, the Netherlands, and Switzerland. Belgium and France are intermediate cases between family and formal based LTC; I group them with formal care countries based on their public LTC spending. The countries with family based care systems include the Mediterranean countries, Germany, and Austria. I group the Czech Republic and Poland, which are not part of Haberkern and Szydlik's study, with the family care countries as both countries have very low public expenditures on LTC. Consistent with the motivation for the group selection, the two country groups also differ with respect to other institutional characteristics related to LTC as well as in their public opinion related to elderly care as shown in Table 1. Compared to formal care countries, family care countries have on average a lower share of public health expenditure, fewer nursing home beds per population in need, less financial support for informal caregivers, and elderly care is generally considered to be the responsibility of the family rather than the state.<sup>7</sup> These institutional characteristics and cultural values have been shaped over a long period of time and are influenced by the political process. However, while the institutional setting is endogenous, changing a country's LTC care system or society's view on elderly care requires time. Consequently, the formal help available to a potential caregiver through the LTC system can

<sup>&</sup>lt;sup>7</sup>A caregiver allowance is a public benefit paid directly to the caregiver (Columbo et al., 2011).

	Public LTC exp. (% of GDP) <sup>a</sup>	Public health exp. $(\% \text{ of total})^b$	Nursing home beds (per 1000 pop $65+)^c$	Caregiver allowance <sup>d</sup>	Elderly care state responsibility <sup><math>e</math></sup>	Old age dep. ratio <sup>f</sup>
Countries	with predominantly	family based LTC				
Poland	0.70	70.28	17.10	0	2.34	19.40
Czechia	0.80	84.18	43.30	0	2.45	23.40
$\operatorname{Spain}$	0.80	73.05	29.30	0	2.86	25.80
Germany	1.40	76.45	50.30	1	2.76	31.20
Greece	1.40	65.05	1.40	0	2.30	29.90
Austria	1.60	67.83	38.70	0	2.93	26.20
Italy	1.90	77.84	16.60	0	2.75	31.60
Average	1.23	73.53	28.10	0.14	2.64	26.79
Countries	with strong formal	Ú.				
Switzerland	2.16	64.89	68.90		3.00	25.30
France	2.20	76.75	51.80	0	3.42	26.60
Belgium	2.30	75.91	70.80	1	3.23	26.40
Netherlands	3.80	85.60	68.40	1	3.58	24.40
Sweden	3.90	81.62	80.40	1	3.74	29.20
Denmark	4.50	85.31	52.30	1	4.14	26.70
Average	3.14	78.35	65.43	0.83	3.52	26.43
$^{a}$ European	Commission (2012) and	d Eurostat; 2010 valu	es. <sup>b</sup> OECD Health Dat	a 2013; 2011	or nearest year avail	able.
<sup>c</sup> OECD Het	alth Data 2012 - Long-	Term Care Data; 200	9; Greece: 2000. <sup>d</sup> Colo:	mbo et al. 20	11. <sup>e</sup> SHARE; Averag	ge of
the SHARE	respondents' answers	to the questions who	- the family (1) or the	state $(5)$ - sh	ould bear the respon	-isi-
bility for he	lp with household chor	es/personal care for	older persons. N=2,181	. <sup>f</sup> Eurostat;	2012 values.	

Table 1: Institutional characteristics and social values

be considered an exogenous factor in an individual's decision whether to provide informal care or not. As formal care may be used to complement or substitute informal care, caregiving may be more strenuous in countries with few formal care options compared to countries with more generous formal care alternatives (Bolin et al., 2008a; Bonsang, 2009).

#### 3 Theoretical framework

This section presents the described interrelation between work, caregiving, cognitive ability, and health in a simple theoretical model to demonstrate how this interrelation may be influenced by the caregiving context. The sole decision maker in my model is the adult daughter who is altruistic towards her elderly parent. The daughter's utility is given by

$$U^D = U^D(C, L, U^P), \tag{1}$$

where C, L, and  $U^P$  denote consumption, leisure, and parental well-being, respectively, while the parent's utility is given by

$$U^P = U^P(CG, OC), (2)$$

where CG represents informal care provided by the daughter, and OC indicates other care or consumption.<sup>8</sup> Hence, the model does not rule out the possibility of formal care or informal care provided by another family member. However, I assume that informal care received from the daughter has a special value to the parent and care provided by someone else is not a perfect substitute. For this reason, I also do not allow for the option of substituting informal caregiving with a monetary transfer to the parent. The daughter's altruism may be rooted in the desire to support her parent but may also arise from family obligations or social and cultural norms to provide care. I assume that both the daughter's and the parent's utility are increasing in each input with decreasing marginal utility.

The daughter's time constraint is given by

$$T = H + L + CG,$$
(3)

where T is the individual's total time endowment and H denotes hours worked. If a daughter wants to increase the amount of time devoted to informal care, she is forced to either reduce her leisure time or

 $<sup>^{8}</sup>$ Since this paper focuses on the daughter as the sole decision maker, I abstract from possible altruism of the parent towards the daughter.

her work hours. Her budget constraint is given by

$$C = Y^* + w(S)H - c(CG), \tag{4}$$

where  $Y^*$  represents outside income, w denotes the wage, which depends on the daughter's stock of health capital and cognitive ability - for ease of notation represented by a composite measure S - with S > 0,  $\partial w(S)/\partial S \ge 0$ , and c(CG) captures the cost of caregiving. This cost can be interpreted as a combination of monetary effort cost and additional expenditures for medication or equipment for the care recipient. A more generous LTC system may reduce the monetary cost component by providing coverage for care related expenditures or by offering financial benefits to caregivers. I assume a quadratic cost of caregiving and set  $c(CG) = \frac{e}{2}CG^2$ .

Lastly, the stock of health capital and cognitive ability evolves according to

$$S = S_0 + \gamma CG,\tag{5}$$

where  $S_0$  represents the initial level of health capital and cognitive ability and  $\gamma$  represent the effectiveness of caregiving as an investment in the production of health capital and cognitive ability (Grossman, 1972).<sup>9</sup> If  $\gamma$  is positive, caregiving increases the stock of health capital and cognitive ability, otherwise caregiving is detrimental to the caregiver's health or cognitive ability.<sup>10</sup>

The daughter chooses the amount of consumption, leisure, work hours, and hours providing care in order to maximize her utility subject to her time and budget constraints and the law of motion for health capital and cognitive ability. Her maximization problem is equivalent to maximizing the Lagrangian:

$$\mathcal{L} = U^D(C, L, U^P(CG, OC))$$

$$+ \lambda_1(Y^* + w(S)H - \frac{e}{2}CG^2 - C)$$

$$+ \lambda_2(T - H - L - CG)$$

$$+ \lambda_3(S - S_0 - \gamma CG).$$
(6)

where  $\lambda_1$  to  $\lambda_3$  are the Lagrange multipliers for the budget constraint, the time constraint and the law of motions for health capital and cognitive ability. Assuming the daughter chooses a strictly positive

<sup>&</sup>lt;sup>9</sup>For simplicity, I abstract from other possible investments and use a combined measure of both health and cognitive ability. Differential effects of caregiving on these outcomes can easily be incorporated by allowing for different coefficients for each outcome.

<sup>&</sup>lt;sup>10</sup>Extending the model to a multi-period set-up or allowing for the depreciation of health capital and cognitive ability will affect the magnitude of health and cognitive benefits derived from caregiving if investments in health or cognitive capital persist (caregiving in one period will increase - or decrease - utility in future periods by increasing - or decreasing the stock of health or cognitive capital) but does not change the qualitative findings of the model.

amount of consumption and leisure and defining  $U_X \equiv \frac{\partial U}{\partial X}$ , an optimal solution to her maximization problem is characterized by the following necessary first-order conditions (FOCs):

$$i) \quad U_C^D - \lambda_1 = 0,$$

$$ii) \quad U_L^D - \lambda_2 = 0,$$

$$iii) \quad w(S)\lambda_1 - \lambda_2 \le 0, \quad H \ge 0,$$

$$iv) \quad U_{UP}^D U_{CG}^P - eCG\lambda_1 - \lambda_2 - \gamma\lambda_3 \le 0, \quad CG \ge 0,$$

$$v) \quad \frac{\partial w(S)}{\partial S} H\lambda_1 + \lambda_3 = 0,$$

where in iii) and iv) at most one inequality can be strict.

After substituting and rearranging terms, the FOCs can be expressed as:

$$w(S)U_C^D \le U_L^D, \quad H \ge 0,\tag{7}$$

which characterizes the typical trade-off between the returns and costs of work from increased consumption and foregone leisure, respectively, and determines whether the individual will work or not, and

$$CG \ge \frac{1}{eU_C^D} \left[ U_{U^P}^D U_{CG}^P + U_C^D \frac{\partial w(S)}{\partial S} H\gamma - U_L^D \right], \quad CG \ge 0,$$
(8)

which determines whether the individual will provide care or not.<sup>11</sup> Again, both inequalities in the same line cannot be strict. For a positive level of care, the daughter equalizes the marginal benefits and costs of providing care. Consequently, the hours of care provided increase in the marginal benefit derived from the parent's well-being  $(U_{U^P}^D)$  times the parent's marginal utility from caregiving  $(U_{CG}^D)$  and decreases in the cost of providing care (e) weighted by the marginal utility of consumption  $(U_C^D)$  and in the marginal utility of leisure  $(U_L^D)$ . Depending on whether caregiving increases  $(\gamma > 0)$  or decreases  $(\gamma < 0)$  health and cognitive ability, caregiving increases or decreases in the marginal benefit of an increase in health and cognitive ability  $(U_C^D \frac{\partial w(S)}{\partial S} H)$ .<sup>12</sup>

#### 4 Empirical strategy

Using  $y_{it}$  to denote individual *i*'s labour market, cognitive, or health outcome of interest at time t and rewriting the FOCs in terms of the outcome, the effect of caregiving can be approximated by the

<sup>&</sup>lt;sup>11</sup>If the daughter is employed, Equations 8 can be explicitly expressed in terms of the wage:  $CG \geq \frac{1}{eU_C^D} \begin{bmatrix} U_U^D & U_C^P & U_C^D & \frac{\partial w(S)}{\partial S} H\gamma - w(S) U_C^D \end{bmatrix}$ 

 $f_2$  An empirical analysis of the relationship between socio-demographic and institutional characteristics and caregiving behaviour is provided in Heger (2014).

linear estimation equation

$$y_{it} = \alpha_1 C G_{it} + \alpha_2 X_{it} + c_i + u_{it}, \tag{9}$$

where  $y_{it}$  is a function of caregiving activity,  $CG_{it}$ , individual demographic and socioeconomic characteristics,  $X_{it}$ , an individual specific error term,  $c_i$ , and an idiosyncratic error term,  $u_{it}$ .

As unobserved time-invariant individual characteristics captured by the individual specific error term  $c_i$  may be correlated with caregiving behaviour, I estimate Equation 9 using fixed effects methods. Fixed effects estimation assumes strict exogeneity of the explanatory variables conditional on the individual fixed effects but allows for arbitrary correlation between the observed independent variables and  $c_i$ , that is  $E(CG'_ic_i) \neq 0$  and  $E(X'_ic_i) \neq 0$  (Wooldridge, 2002). Allowing for unobserved, time-invariant heterogeneity in this way appropriately describes the data because individual characteristics and personality traits, such as closeness to the parent, selflessness, or upbringing, may influence caregiving behaviour but can reasonably be assumed to be constant over time for individuals aged 50 and above. However, time-varying endogeneity may still be an issue. For example, as work and caregiving activities both compete for the individual's time endowment, an unobserved shock may change the daughter's time cost and impact both labour supply and caregiving activities.

To account for such time-varying endogeneity, I use an indicator for whether at least one parent suffers from poor health as an instrument for caregiving behaviour.<sup>13</sup> Poor parental health increases the parent's need for care and thus encourages the daughter to provide care by increasing the benefits of care provision as depicted by Equation 8. Likewise, an improvement of the parent's health status or the death of the parent reduces or eliminates the need for informal care.<sup>14</sup> In order for parental health (*PH*) to be a valid instrument, parental health must influence the daughter's outcome under consideration only by affecting caregiving behaviour. Thus, in addition to  $corr(CG_{it}, PH_{it}) \neq 0$ ,  $E(u_{it}|PH_{it}) = 0$  must hold. While the level of parental health does not represent a valid instrument, fixed effects estimation only considers within individual variation, which means a change in parental health serves as an instrument for a change in caregiving behaviour. Arguably, after controlling for all other covariates, a worsening of the parent's health status only influences the adult daughter's labour supply, cognitive ability or health through its effect on caregiving behaviour.<sup>15</sup> Since the goal is to estimate separate effects for family

<sup>&</sup>lt;sup>13</sup>The measure is based on the daughter's assessment of her mother's and father's health. The possible categories for parental health changed from "very good", "good", "fair", "poor", and "very poor" in wave 1 to "excellent", "very good", "good", "fair", and "poor" in waves 2 and 4. Jürges et al. (2008) show that it is still possible to obtain a consistent measure by combining the categories "poor" and "very poor" and "very good" and "excellent". Poor parental health indicates poor health based on the newly constructed health variable. If two parents are alive parental health is defined as the health status of the parent in worse health.

<sup>&</sup>lt;sup>14</sup>While self-perceived individual health is not necessarily a good objective measure of individual health, the daughter's decision to provide care depends on her perception of her parent's health rather than on the parent's objective health level. Hence, potential subjectivity of the measure is less of a concern in this study. Besides, systematic misreporting of parental health does not appear to be a problem (see Appendix A for a further discussion).

 $<sup>^{15}</sup>$ While this argument has also been made by Van Houtven et al. (2013) in their analysis of the effect of caregiving on work for mature men and women in the U.S., Bobinac et al. (2010) find a direct health effect of the declining health of a

and formal care countries, I use a change in parental health interacted with the indicators for each country group as instruments, as a change in parental health may have a differential effect on daughters' caregiving decision based on the availability of formal care alternatives.<sup>16</sup>

Which variables are included in  $X_{it}$  depends on the outcome of interest. In all regressions I control for the number of chronic conditions, limitations with activities of daily living (ADL), and instrumental ADL (IADL). Changes in these health measures capture relatively severe changes in health and I treat them as exogenous since it is unlikely that caregiving impacts these measures within the observed time horizon. Since the death of or separation from a partner may impact the outcome under consideration, I control for whether a person is married or living in a registered partnership and household size. To capture changes over time, I control for a linear and quadratic effect of the respondent's age. Time-invariant covariates cannot be included in fixed effects estimation as they are part of the individual specific fixed effects. Hence, educational attainment cannot be used as explanatory variables since educational levels usually no longer change for individuals over the age of 50. However, life expectancy, population health, and the general education level differ across countries. As a result, the natural deterioration rate of cognitive ability or health might also differ across countries. To capture this effect, I include country dummies interacted with age (both linear and quadratic terms). To pick up additional variation over time due to a change in the economic situation or as respondents become familiar with the type of cognitive skills questions, I include two wave dummies. Finally, I control for household net wealth by including dummies for country specific wealth quartiles.

For labour market outcomes, I also control for being above the country specific official retirement age.<sup>17</sup> As few individuals actually work until the official retirement age, I further include two indicators for being within two and within five years of the country specific official retirement age.

Labour market participation may in turn affect cognitive ability and health by providing cognitive stimulation or affecting physical activity and stress levels. When analyzing cognitive and health outcomes, I control for whether a person is employed (including self-employed). Similarly, social participation may also serve as an investment in cognitive ability or health. I capture such activities by an indicator variable which equals one if the respondent participated in any of the following activities: voluntary or charity work; attended an educational or training course; gone to a sport, social or other kind of club; taken part in activities of a religious organization; or taken part in a political or community-related organization. Moreover, when studying daughters' mental health, I include an indicator for parental loss, which equals one if a parent passed away since the last interview. As I focus on mature daughters it

close relative. I provide further evidence of the suitability of the instrument in Subsection 6.1.

<sup>&</sup>lt;sup>16</sup>Estimation is performed using the Stata command xtivreg2 (Schaffer, 2012).

 $<sup>^{17}</sup>$ Information for country specific retirement ages is based on the official retirement age in 2010 reported by the OECD. Country specific retirement ages are listed in Table 8 in the Appendix.

seems reasonable to assume that the declining health of a parent is seen as a natural occurrence, which does not directly affect the daughter's mental health after controlling for all other covariates including the loss of a parent.<sup>18</sup>

#### 5 Data

The data comes from wave 1, 2 and 4 of the Survey for Health, Ageing and Retirement in Europe (SHARE) collected in 2004/2005, 2006/2007, and 2011/2012 covering the population 50+ in Europe.<sup>19</sup> SHARE is the first data set to include information on health, cognitive ability, socio-economic status, and social engagement of the elderly at a pan-European level, which provides the unique opportunity to study the effect of institutional differences.<sup>20</sup> I limit my sample to women aged 50 to 70 who participated in two or more interviews and who have at least one living parent at the time of their first interview.<sup>21</sup> Countries participating in all three waves are Austria, Germany, Sweden, the Netherlands, Spain, Italy, France, Denmark, Switzerland, and Belgium. Greece participated in wave 1 and 2; Czechia and Poland are part of SHARE since wave 2. After deleting observations with missing information, my sample includes 3,309 individuals or 7,591 person-wave observations.<sup>22</sup>

Sample size and caregiving rates by country and country group are shown in Table 2.<sup>23</sup> In addition, Table 9 in the Appendix shows changes in caregiving activities over time. Infrequent caregiving is more common in formal care countries, whereas daily caregiving is more common in family care countries.<sup>24</sup> Descriptive statistics for all further variables are presented in Table 3 for the total sample and separately for individuals who never provide care (never caregivers) and individuals who state they are caregivers in at least one interview (ever caregivers). Ever caregivers tend to be healthier, have higher cognitive test scores, and are more educated, underlining the need to account for the endogeneity of caregiving. Not surprisingly, ever caregivers also have more frequently a parent in poor health.

<sup>&</sup>lt;sup>18</sup>Also see the discussion in Subsection 6.1.

 $<sup>^{19} \</sup>rm Wave~3$  of SHARE consists of a special retrospective survey (SHARELIFE), which does not ask about caregiving activities. The SHARE questionnaires and data are available at www.share-project.org.

<sup>&</sup>lt;sup>20</sup>Data is collected using a computer-assisted personal interviewing technique (CAPI). Sample selection varies across countries from simple random selection of households to multi-stage designs due to varying institutional conditions regarding sampling (Börsch-Supan and Jürges, 2005, Ch.5). All results presented in this paper are weighted using population weights based on individuals' first observation.

<sup>&</sup>lt;sup>21</sup>For older individuals, the labour market outcomes are largely irrelevant, the importance of investments in health and human capital is declining, and the number of respondents with a living parent is very small.

 $<sup>^{22}</sup>$ SHARE provides five imputed data sets to deal with missing information in key variables (see Christelis (2011) for details). The results in this study are based on the first of these data sets. Item non-response is minor for non-financial questions and the results are not sensitive to which imputed data set is used

<sup>&</sup>lt;sup>23</sup>Caregiving is measured by an indicator variable for whether a daughter provided care during the 12 months prior to the interview and includes help with personal care (e.g. dressing, bathing) or practical household help (e.g. help with home repairs, shopping, household chores) provided outside or insight the household. In wave 2, respondents who have been interviewed in wave 1 are asked whether they have provided care since the last interview, which corresponds to approximately 24 months. To distinguish between help during a short-term sickness, a daughter providing help to a co-residing parent is only considered a caregiver if this help occurs almost daily during at least three months.

<sup>&</sup>lt;sup>24</sup>Equality of caregiving rates in all countries can be rejected with a p-value of 0.000 for all frequencies of care.

		Caregiv	ing rates	
Country	Sample size	Any	Weekly	Daily
Poland	112	0.20	0.16	0.11
Italy	285	0.26	0.22	0.14
Austria	132	0.27	0.21	0.12
Greece	275	0.27	0.21	0.13
Spain	213	0.29	0.20	0.18
France	404	0.29	0.17	0.07
Switzerland	175	0.37	0.20	0.06
Germany	276	0.40	0.25	0.08
Denmark	247	0.40	0.19	0.02
Czechia	113	0.43	0.27	0.05
Sweden	377	0.44	0.17	0.03
Netherlands	312	0.45	0.29	0.05
Belgium	388	0.46	0.35	0.14
Total	3,309	0.33	0.21	0.10
Family care countries	1,406	0.31	0.22	0.12
Formal care countries	1,903	0.35	0.20	0.07

Table 2: Sample size and caregiving rates by country

Countries ordered by caregiving rates (any frequency).

Weighted values based on individuals' first observation.

#### Outcome variables

Labour force participation is measured by an indicator for whether the daughter is employed or self-employed and the total number of hours a respondent worked in her main or secondary job per week.

Next, I use four measures of cognitive ability. First, verbal fluency is measured by the number of animals the respondent can name. Each acceptable animal counts as one point up to a maximum score of 100. Second, short term word recall is tested by the respondent's ability to remember a list of ten words the interviewer reads to her. The score is given by the number of correctly recalled words. At the end of the cognitive function module, the respondent is asked to state as many of the words as she remembers from the list read to her earlier to measure long term word recall. The score is again given by the number of correctly recalled words. Finally, numeracy skills are assessed based on a series of four math questions taken from everyday life situations. The numeracy score ranges from 1 (poor numeracy skills) to 5 (good numeracy skills).<sup>25</sup>

Furthermore, I consider both mental and physical health outcomes of caregivers. Mental health is measured by the EURO-D depression scale ranging from 0 (not depressed) to 12 (very depressed). This measure has been developed to capture symptoms of depression (including, for example, feelings of guilt, loss of interest, trouble sleeping, fatigue, or tearfulness).<sup>26</sup> Besides, I analyze changes in self-

 $<sup>^{25}</sup>$ SHARE also includes a fifth measure of cognitive ability, which tests respondents' knowledge of the year, month, and day of the interview. However, most respondents perform well on this test and the measure includes too little variation for an informative analysis. The time limit for each cognitive test is one minute.

 $<sup>^{26}\</sup>mathrm{For}$  more information see Castro-Costa et al. (2008).

	Never car	regivers	Ever car	egivers		
	Mean	Ν	Mean	Ν	Diff.	Sig.
Labour force participation						
Employed	0.47	1,649	0.51	1,653	0.04	
Weekly hours worked (if hours>0)	34.82	806	34.54	916	-0.28	
Cognitive ability						
Verbal fluency	19.73	$1,\!636$	21.85	$1,\!649$	2.12	1%
Short-term word recall	5.31	1,641	5.71	1,652	0.40	1%
Long-term word recall	3.90	1,642	4.40	1,652	0.50	1%
Numeracy	3.29	$1,\!651$	3.48	$1,\!654$	0.19	1%
Health						
Depression (EURO-D)	2.91	1,631	2.67	1,648	-0.24	$5^{\circ}$
Self perceived health	2.92	1,653	2.69	1,656	-0.23	1%
Grip strength	28.99	1,551	30.15	$1,\!617$	1.16	$1^{\%}$
Socioeconomic and demograph	ic inform	ation				
Age	55.59	1.653	55.67	1.656	0.08	
Age squared/100	31.09	1,653	31.16	1.656	0.07	
Married	0.72	1,653	0.72	1.656	0.00	
Household size	2.44	1,653	2.40	1.656	-0.04	
Number of chronic conditions	1.20	1,653	1.10	1,656	-0.10	$10^{\circ}$
Number of limitatins with ADL	0.10	1,653	0.03	1,656	-0.07	1%
Number of limitatins with IADL	0.18	1,653	0.08	1,656	-0.10	1%
Other activities	0.32	1,642	0.46	1,651	0.14	1%
Reached off, retirement age	0.13	1,653	0.11	1,656	-0.02	
2 years to off, retirement age	0.21	1,653	0.20	1,656	-0.01	
5 years to off, retirement age	0.40	1,653	0.37	1,656	-0.03	
Household net worth	320,084	1,653	364,184	1,656	44,100	
Primary education or less	0.25	1,649	0.16	$1,\!653$	-0.09	1%
Secondary education	0.51	1,649	0.52	$1,\!653$	0.01	
Post-secondary education	0.24	1,649	0.32	$1,\!653$	0.08	1%
Parental information						
Parent in poor health	0.27	1,573	0.34	1,593	0.07	1%
Parent in fair health	0.36	1,573	0.36	1,593	0.00	
Parent in good health	0.30	1,573	0.23	1,593	-0.07	1%
Parent in very good health	0.07	1,573	0.07	1,593	0.00	

#### Table 3: Descriptive statistics based on first observations

Weighted values based on individuals' first observation. N=3,309

perceived general health based on the categories excellent (1), very good (2), good (3), fair (4), and poor (5). While self-perceived health is a strong predictor of future morbidity and mortality (Idler and Benyamini, 1997; Maddox and Douglass, 1973), respondents' perception of their own may change when faced with the declining health of a parent. To complement self-perceived health, I use grip strength as an objective indicator of physical health (Ziebarth, 2010; Jürges, 2007). Grip strength is measured in a gripping exercise where respondents are asked to squeeze their hand as hard as they can for a couple of seconds.<sup>2728</sup>

#### 6 Results

This section presents the estimation results for the effect of caregiving on labour force participation, cognitive ability, and health in turn. Results are presented for the fixed effects estimation with parental health as an instrument for caregiving activities (FE-IV) and without (FE); the full regression results and first stage regressions are shown in the Appendix.<sup>29</sup> I test whether individual fixed effects are needed to account for time constant endogeneity by comparing OLS-IV and FE-IV estimates (Fletcher and Lehrer, 2009). Tests of joint significance of the individual specific fixed effects are statistically significant at the 1% significance level for all specifications. Since for caregiving in family (formal) care countries only having a parent in poor health interacted with the dummy for family (formal) care country is informative, I report Angrist-Pischke F tests for weak instruments, which test weak identification of one instrument while partialling out the effect of the other instrument (Angrist and Pischke, 2009, pp. 217-18, Baum et al., 2010). Critical values are not available, but the critical values for the Cragg-Donald F statistic with one endogenous variable reported by Stock and Yogo (2005) can be used as reference.<sup>30</sup> Weak instruments can be rejected for all but one specification: When hours worked is the dependent variable, poor parental health is not a strong predictor for caregiving behaviour.<sup>31</sup>

Table 4 shows the estimated effects of caregiving on employment and hours worked in family and formal care countries. The FE regression results show a slight increase in employment of 4 percentage points for weekly and any frequency of caregiving in formal care countries but no effect on hours. However, individuals may self-select into caregiving if their job allows them to combine work and caregiving. For

 $<sup>^{27}\</sup>mathrm{Two}$  measures are taken for each hand; I use the maximum result of the four measures.

 $<sup>^{28}</sup>$ The measures of cognitive ability, self-perceived health, and grip strength show a progressive decline with age, underlining the importance of efforts to increase or preserve these outcomes in later life (Heger, 2014).

<sup>&</sup>lt;sup>29</sup>If exogeneity of caregiving cannot be rejected, both the FE and FE-IV estimates can be interpreted as causal effects. Results can differ due to the smaller sample size when using the instrument and because the FE-IV specification captures the effect of caregiving as a result of poor parental health. Such caregiving is likely to be more physical or emotionally demanding.

<sup>&</sup>lt;sup>30</sup>The critical values as computed by Stock and Yogo (2005) for 10, 15, 20, and 25% maximal IV size bias in Wald tests are 16.38, 8.96, 6.66, and 5.53, respectively (Stock and Yogo, 2005).

<sup>&</sup>lt;sup>31</sup>Since FE-IV is very data hungry, this result is not too surprising given the reduced sample size since only individuals who work positive hours are included. In addition, working individuals may be less likely to change their caregiving behaviour in response to a change in parental health than non-working individuals.

		FE			FE-IV	
Frequency of care:	Any	Weekly	Daily	Any	Weekly	Daily
			En	ployed		
Caregiver	-0.028	-0.013	-0.016	-0.344**	-0.370**	-0.600**
(family care country)	(0.025)	(0.027)	(0.030)	(0.144)	(0.168)	(0.272)
Caregiver	$0.042^{**}$	$0.042^{*}$	-0.021	0.018	0.023	0.038
(formal care country)	(0.020)	(0.024)	(0.037)	(0.149)	(0.194)	(0.257)
Equality of effects (p-value)	0.030	0.133	0.931	0.082	0.125	0.090
Exogeneity (p-value)				0.041	0.037	0.040
AP F-Stat (family)				34.60	30.07	18.92
AP F-Stat (formal)				20.52	18.46	21.75
			Hour	s worked		
Caregiver	2.032	0.911	1.714	2.474	1.928	2.054
(family care country)	(1.482)	(1.506)	(1.602)	(15.060)	(10.637)	(11.609)
Caregiver	0.441	0.054	0.082	2.480	5.585	5.354
(formal care country)	(0.845)	(1.095)	(2.266)	(6.748)	(15.726)	(14.489)
Equality of effects (p-value)	0.353	0.647	0.566	1.000	0.850	0.858
Exogeneity (p-value)				0.966	0.939	0.937
AP F-Stat (family)				2.05	4.51	7.25
AP F-Stat (formal)				11.39	2.76	6.68
Standard errors clustered at	household	level in pa	arentheses.	*** p<0.01,	** p<0.05	* p<0.1

Table 4: The effect of informal caregiving on labour force participation

employment, exogeneity of caregiving is rejected at the 5% level. The FE-IV regressions estimate the effect of caregiving when caregiving is triggered by a decline in parental health and thus account for the endogeneity of caregiving. When the daughter responds to the parent's need for care, any frequency, weekly, and daily caregiving reduces the probability that the daughter is employed by 34, 37, and 60 percentage points, respectively, in family care countries, while caregiving in formal care countries has no effect on caregivers' employment probability. The large effect found for family care countries and the absence of an effect in formal care countries is in line with the finding by Crespo and Mira (2010) who find a 50% drop in the employment probability in Southern European countries and no change for Northern European countries but contradicts studies that find no or only negligible effects in all countries as well as studies that report negative effects in Northern European countries.<sup>32</sup> As mentioned above, the FE-IV regressions for hours worked suffer from weak instrument problems and results are only shown for completeness. Bolin et al. (2008b) cannot reject time varying exogeneity of hours worked with respect to caregiving analyzing similar countries, in which case the FE estimation results can be interpreted as causal effects.<sup>33</sup>

Table 5 shows the results for verbal fluency, short- and long-term word recall, and numeracy skills.

 $<sup>^{32} \</sup>rm While$  the distinction between family and formal care countries is not based on geography, in general, Northern countries generally belong to formal care countries, while Southern and Eastern countries are family care countries.

 $<sup>^{33}</sup>$ However, Van Houtven et al. (2013) reject exogeneity for the U.S., thus the result that caregiving does not affect hours worked should be interpreted with some caution.

		FE			FE-IV	
Frequency of care:	Any	Weekly	Daily	Any	Weekly	Daily
			Verba	l fluency		
Caregiver	$1.136^{***}$	0.845**	$1.158^{**}$	3.739**	4.145*	$6.405^{*}$
(family care country)	(0.389)	(0.425)	(0.504)	(1.873)	(2.149)	(3.478)
Caregiver	-0.235	0.329	-0.503	-1.132	-1.425	-1.999
(formal care country)	(0.339)	(0.421)	(0.771)	(2.340)	(2.992)	(4.051)
Equality of effects (p-value)	0.007	0.383	0.069	0.111	0.135	0.116
Exogeneity (p-value)				0.383	0.265	0.259
AP F-Stat (family)				34.30	28.79	19.29
AP F-Stat (formal)				21.02	20.24	21.16
		C		1 11		
Conorinon	0.049	0.050	O 145	0 252	0.286	0.592
(family and accenture)	(0.042)	-0.000	-0.140	-0.332	-0.380	-0.005
(family care country)	(0.111)	(0.120)	(0.152)	(0.303)	(0.022)	(0.985)
(formal comp country)	0.095	(0.214)	(0.124)	(0.452)	(0.917)	(1, 1, 40)
(Iormai care country)	(0.088)	(0.095)	(0.134)	(0.058)	(0.817)	(1.140)
Equality of effects (p-value)	0.719	0.095	0.025	0.500	0.551	0.508
AD E Chat (famil)				0.059	0.703	0.804
AP F-Stat (family)				34.20	28.72	19.22
AP F-Stat (formal)				21.82	20.39	21.51
		$\mathbf{L}$	ong-tern	ı word recal	1	
Caregiver	-0.132	-0.232*	-0.264*	-0.286	-0.330	-0.545
(family care country)	(0.109)	(0.125)	(0.136)	(0.608)	(0.678)	(1.055)
Caregiver	0.120	0.168*	0.331*	-0.856	-1.097	-1.502
(formal care country)	(0.090)	(0.092)	(0.189)	(0.688)	(0.904)	(1.224)
Equality of effects (p-value)	0.074	0.010	0.010	0.540	0.499	0.552
Exogeneity (p-value)				0.330	0.343	0.294
AP F-Stat (family)				34.20	28.72	19.22
AP F-Stat (formal)				21.82	20.39	21.51
			Nur	neracy		
Caregiver	0.014	0.034	-0.070	-0.411	-0.457	-0.708
(family care country)	(0.044)	(0.048)	(0.062)	(0.262)	(0.296)	(0.456)
Caregiver	-0.012	-0.029	-0.014	0.060	0.073	0.106
(formal care country)	(0.033)	(0.035)	(0.055)	(0.270)	(0.333)	(0.456)
Equality of effects (p-value)	0.647	0.299	0.500	0.219	0.240	0.209
Exogeneity (p-value)				0.212	0.179	0.289
AP F-Stat (family)				34.04	28.63	19.15
AP F-Stat (formal)				20.73	20.55	21.65
Standard errors clustered at	household l	evel in par	entheses.	*** p<0.01, '	** p<0.05,	* p<0.1

Table 5: The effect of informal caregiving on cognitive ability

Exogeneity of caregiving cannot be rejected for any cognitive measure. I find caregiving increases individuals' verbal fluency scores by approximately one additional word in family care countries but has no statistically significant effect in formal care countries. Caregiving initiated by poor parental health leads to even larger increases in verbal fluency scores in family care countries (4 to 6 additional words for any frequency of caregiving and daily caregiving, respectively), though standard errors also increase. While the effects in formal care countries remain statistically insignificant, the point estimates are consistently negative.

Short-term word recall is considered a more difficult cognitive skill than verbal fluency. I find significant positive effects for weekly and daily caregiving in formal care countries in the FE specification of 0.21 and 0.28 words, respectively. Caregiving induced by a change in parental health does not lead to statistically significant effects. However, the point estimates are negative in family care countries and positive in formal care countries and increase (in absolute terms) with the frequency of care.

Long-term word recall and numeracy are even more difficult cognitive skills. Like short-term word recall, weekly and daily caregiving increases long-term word recall in formal care countries in the FE specification by 0.23 and 0.26 words, respectively. Contrary, in family care countries caregiving decreases long-term word recall by 0.17 and 0.33 words for weekly and daily caregiving, respectively. The effects are significant at the 10% level. Unlike before, however, the point estimates in the FE-IV specification are negative for both family and formal care countries, though not statistically significant. No significant effects are found for numeracy skills. Possibly, the stress of caregiving has more negative consequences for difficult cognitive skills (Arpino and Bordone, 2012).

Results for health outcomes are reported in Table 6. Caregiving does not affect mental health in the FE specification. However, caregiving in response to a parent's declining health leads to between 2 and 4 additional depressive symptoms; the effects are not significantly different between family and formal care countries. Caring for a parent in poor health is arguably much more emotionally demanding than general caregiving, which can explain the large difference between the FE and IV-FE results.

I find a slight improvement in self-perceived health as a result of weekly and any frequency of caregiving in formal care countries using the FE specification. The point estimates in family care countries as well as all estimates using the FE-IV specification are positive, that is they suggest a detrimental effect of caregiving on self-perceived health, but are statistically insignificant.

The beneficial health effects of caregiving in formal care countries, however, are not mirrored by the objective grip strength measure, which casts doubt on the reliability of self-perceived health as a valid health measure in this context because caregiving may change caregivers' point of reference for good health. Weekly caregiving decreases grip strength in formal care countries by 0.50 kg, whereas in family care countries, caregiving increases grip strength by 0.67 kg and 0.65 kg for weekly and any frequency

		FE			FE-IV				
Frequency of care:	Any	Weekly	Daily	Any	Weekly	Daily			
			EUR	lO-D					
Caregiver	0.202	0.204	0.115	1.849**	$1.935^{*}$	$2.931^{*}$			
(family care country)	(0.133)	(0.150)	(0.174)	(0.935)	(1.022)	(1.610)			
Caregiver	0.104	-0.014	0.328	$2.585^{**}$	3.179**	3.783**			
(formal care country)	(0.110)	(0.126)	(0.313)	(1.070)	(1.278)	(1.547)			
Equality of effects (p-value)	0.568	0.265	0.551	0.596	0.441	0.715			
Exogeneity (p-value)				0.003	0.002	0.004			
AP F-Stat (family)				31.12	28.17	19.93			
AP F-Stat (formal)				19.54	18.72	21.78			
		S	elf-percei	ved health					
Caregiver	0.002	0.035	0.069	0.223	0.244	0.410			
(family care country)	(0.053)	(0.058)	(0.065)	(0.296)	(0.319)	(0.522)			
Caregiver	$-0.109^{**}$	$-0.111^{**}$	-0.133	0.407	0.522	0.713			
(formal care country)	(0.043)	(0.051)	(0.086)	(0.398)	(0.509)	(0.705)			
Equality of effects (p-value)	0.102	0.058	0.061	0.715	0.647	0.730			
Exogeneity (p-value)				0.293	0.353	0.352			
AP F-Stat (family)				34.32	30.08	19.07			
AP F-Stat (formal)				21.80	20.38	21.51			
			Cuin	tu o out lo					
Constitute	0 646**	0 667**	0.224	1 019	0 101	2.405			
(family and family and	(0.040)	(0.007)	(0.324)	1.912	(1.740)	(0.007)			
(family care country)	(0.306)	(0.335)	(0.304)	(1.533)	(1.(42))	(2.927)			
Caregiver	-0.343	-0.499*	-0.302	-2.246	-2.708	-3.650			
(formal care country)	(0.243)	(0.289)	(0.445)	(1.560)	(1.822)	(2.465)			
Equality of effects (p-value)	0.011	0.008	0.061	0.058	0.066				
Exogeneity (p-value)				0.322 0.357 0.218					
AP F-Stat (family)				31.68	26.04	15.77			
AP F-Stat (formal)				18.67	19.58	22.14			
Standard errors clustered at	household l	level in par	entheses. *	** p<0.01,	** p<0.05.	* p<0.1			

#### Table 6: The effect of informal caregiving on health

of caregiving. Using the FE-IV specification leads to larger positive and larger negative effects if family and formal care countries, respectively, but the effects are statistically insignificant.

#### 6.1 Robustness checks

In this subsection, I test the sensitivity of my results with respect to the choices and assumptions regarding covariates and instruments. I provide different robustness checks for the various outcomes of interest. For labour force participation, I re-estimate the model without the dummies for wealth quartiles due to possible endogeneity concerns. Excluding these covariates does not change the results. Likewise, replacing the dummies for wealth quartiles by an indicator for positive income from other household members also does not alter my findings. Results are available upon request.

Next, since being retired might change the rate of cognitive decline or affect health outcomes, I add dummies for having been retired in t-1 and, when observed, in t-2 to the models for cognitive ability and health outcomes. Again, the findings remain the same; results are available upon request.

Further, the exclusion restriction of my instrument, which relies on the assumption that a decline in parental health does not directly affect the daughter's outcome variable after controlling for all other covariates, is arguably strongest with respect to mental health, where a direct effect might seem plausible (Bobinac et al., 2010). To address this concern, I present estimates of the mental health outcome using a dummy for having only one living parent as an instrument, which allows me to use poor parental health as explanatory variable. Only having one parent increases the probability of becoming a caregiver since the living parent's spouse can no longer provide care, but it should not have a direct effect on the adult daughter's mental health (note that I still control for parental loss). Unfortunately, having a single parent is not an overly strong predictor of caregiving behaviour in family care countries nor of frequent caregiving in formal care countries. Thus I present estimates for any frequency of caregiving only, both for family and formal care countries combined and separately for formal care countries (see Table 23 in the Appendix). The negative effect of caregiving is no longer significant in the combined sample. Since caregiving to a single parent might not be as stressful as caregiving to a parent in poor health, this finding does not contradict my earlier results. In formal care countries, caregiving increases the EURO-D score by 1.7; the effect is statistically significant at the 10% level. Importantly, only the effect of losing a parent but not the effect of having a parent in poor health is significant in the combined sample and neither effect is significant in formal care countries, which supports the use of poor parental health as an instrument.

#### 7 Discussion and conclusion

While care receivers as well as policymakers concerned about LTC expenditures usually prefer informal care provided by a family member, care provision requires caregivers to reduce leisure time or work time and exposes them to additional stress, which may affect their work and well-being. This paper shows the breadth of the effects of parental caregiving on mature daughters by studying labour force participation, cognitive ability, and health outcomes in thirteen European countries and shows that caregiving can lead to both positive and negative effects.<sup>34</sup> To better understand the role of LTC institutions as an important determinant of whether positive or negative effects prevail, I estimate the effects of caregiving in countries with strong formal LTC and countries with predominantly family based LTC. My results show that the effects of caregiving can differ drastically between these groups. Moreover, the frequency of care and whether care is induced by a change in parental health affect the severity of the effects. Contradictory effects of caregiving reported in earlier studies might therefore be driven by institutional differences and by differences in the definition of caregiving.

My results have important implications for the design of LTC systems. For one, they suggest that policies aimed at increasing informal caregiving rates will likely lead to a further reduction in employment rates for mature women in family care countries and additional formal care options could increase the compatibility of work and caregiving obligations in these countries. For another, an important policy goal in both family and formal care countries would be to reduce the mental burden of caregiving. Since mental health problems may reduce the ability of informal caregivers to provide adequate care, such policies would not only help caregivers but would also benefit care receivers. Besides, the positive effects of caregiving should also be noted. Caregiving can increase health and cognitive ability similar to other forms of social engagement. While the institutional context appears to be an influencing factor, exploring the conditions under which caregiving leads to such positive effects in detail deserves further attention.

To conclude, some caveats are in order. First, due to the limited sample size, this study focuses exclusively on daughters. While women tend to provide care more often than men, a growing number of men are becoming caregivers as the traditional gender roles soften and extending the analysis to sons hence represents a natural extension of this study. In particular, the social pressure to provide informal care may be weaker for men but support from co-workers or supervisors may also be weaker, which could provide additional insight about the influence of the cultural and institutional caregiving context on the effects of caregiving. Second, this study can only address short term effects of caregiving, that is it analyzes the change in outcomes after a person starts or ends to provide care and the duration of caregiving is not considered. While these short run effects are greatly relevant to the individual

 $<sup>^{34}</sup>$ Allowing for feedback effects between the health or cognitive outcomes and employment would be a very interesting exercise but is beyond the scope of this study.

caregivers, it would be also extremely interesting to see how caregivers are affected in the long run both from an individual and a policy perspective.

#### References

- Aneshensel, C. S., L.I. Pearlin, J. T. Mullan, S. H. Zarit, and C. Whitlatch (1995) Profiles in caregiving: The unexpected carer (Academic Press, San Diego)
- Angrist, J.D., and J.-S. Pischke (2009) Mostly Harmless Econometrics: An Empiricist's Companion (Princeton University Press)
- Arpino, B., and V. Bordone (2012) 'Does grandparenting pay off? The effect of childcare on grandparents' cognitive functioning.' European Demographic Research Papers No. 4
- Bass, D. M., L. S. Noelker, and L. R. Rechlin (1996) 'The moderating influence of service use on negative caregiving consequences.' The Journals of Gerontology Series B: Psychological Sciences and Social Sciences 51(3), 121–131
- Baum, C. F., M. E. Schaffer, and S. Stillman (2010) 'ivreg2: Stata module for extended instrumental variables/2SLS, GMM and AC/HAC, LIML and k-class regression.' http://ideas.repec.org/c/boc/bocode/s425401.html
- Bertrand, R. M., J. S. Saczynski, C. Mezzacappa, M. Hulse, K. Ensrud, and L. Fredman (2012) 'Caregiving and Cognitive Function in Older Women: Evidence for the Healthy Caregiver Hypothesis.' *Journal of* Aging and Health 24(1), 48–66
- Bobinac, A., N. van Exel, F. Rutten, and W. Brouwer (2010) 'Caring for and caring about: disentangling the caregiver effect and the family effect.' *Journal of Health Economics* 29(4), 549–556
- Bolin, K., B. Lindgren, and P. Lundborg (2008a) 'Informal and formal care among single-living elderly in Europe.' *Health Economics* 17(3), 393–409
- (2008b) 'Your next of kin or your own career?: Caring and working among the 50+ of Europe.' Journal of Health Economics 27(3), 718–738
- Bonsang, E. (2009) 'Does informal care from children to their elderly parents substitute for formal care in Europe?' Journal of Health Economics 28(1), 143–154
- Börsch-Supan, A., and H. Jürges (eds) (2005) The Survey of Health, Ageing and Retirement in Europe -Methodology (Mannheim: MEA)
- Brown, S. L., D. M. Smith, R. Schulz, M. U. Kabeto, P. A. Ubel, M. Poulin, J. Yi, C. Kim, and K. M. Langa (2009) 'Caregiving behavior is associated with decreased mortality risk.' *Psychological science* 20(4), 488–494
- Carmichael, F., and S. A. Charles (1998) 'The labour market costs of community care.' Journal of Health Economics 17(6), 747–765
- (2003) 'The opportunity costs of informal care: does gender matter?' Journal of Health Economics 22(5), 781–803
- Carmichael, F., S. A. Charles, and C. Hulme (2010) 'Who will care? Employment participation and willingness to supply informal care.' *Journal of Health Economics* 29(1), 182–190
- Casado-Marín, D., P. García-Gómez, and Á. López-Nicolás (2011) 'Informal care and labour force participation among middle-aged women in Spain.' SERIEs 2(1), 1–29
- Castro-Costa, E., M. Dewey, R. Stewart, S. Banerjee, F. Huppert, C. Mendonca-Lima, C. Bula, F. Reisches, J. Wancata, K. Ritchie et al. (2008) 'Ascertaining late-life depressive symptoms in Europe: an evaluation of the survey version of the EURO-D scale in 10 nations. The SHARE project.' International Journal of Methods in Psychiatric Research 17(1), 12–29
- Caswell, L. W., P. P. Vitaliano, K. L. Croyle, J. M. Scanlan, J. Zhang, and A. Daruwala (2003) 'Negative associations of chronic stress and cognitive performance in older adult spouse caregivers.' *Experimental* aging research 29(3), 303–318

Christelis, D. (2011) 'Imputation of Missing Data in Waves 1 and 2 of SHARE.' Available at SSRN 1788248

- Ciani, E. (2012) 'Informal adult care and caregivers' employment in Europe.' Labour Economics 19(2), 155–164
- Coe, N. B., and C. H. Van Houtven (2009) 'Caring for mom and neglecting yourself? The health effects of caring for an elderly parent.' *Health Economics* 18(9), 991–1010
- Colombo, F., A. Llena-Nozal, J. Mercier, and F. Tjadens (2011) 'Help Wanted? Providing and Paying for Long-Term Care.' OECD Health Policy Studies
- Crespo, L. (2006) 'Caring for parents and employment status of European mid-life women.' Documentos de Trabajo (CEMFI) (15), 1
- Crespo, L., and P. Mira (2010) 'Caregiving to elderly parents and employment status of European mature women.' Documentos de Trabajo (CEMFI) (7), 1
- de Vugt, M. E., J. Jolles, L. van Osch, F. Stevens, P. Aalten, R. Lousberg, and F. R. J. Verhey (2006) 'Cognitive functioning in spousal caregivers of dementia patients: findings from the prospective MAASBED study.' Age and Ageing 35(2), 160–166
- Dentinger, E., and M. Clarkberg (2002) 'Informal caregiving and retirement timing among men and women gender and caregiving relationships in late midlife.' Journal of Family Issues 23(7), 857–879
- Dobrescu, L., and D. Christelis (2012) 'The Impact of Social Activities on Cognitive Ageing: Evidence From Eleven European Countries.' Working Papers 201207, ARC Centre of Excellence in Population Ageing Research (CEPAR)
- Engelhardt, H., I. Buber, V. Skirbekk, and A. Prskawetz (2010) 'Social involvement, behavioural risks and cognitive functioning among older people.' Ageing and Society 30(5), 779
- Ettner, S. L. (1995) 'The Impact of 'Parent Care' on Female Labor Supply Decisions.' Demography 32, 63-80
- (1996) 'The opportunity costs of elder care.' Journal of Human Resources pp. 189–205
- European Commission (2012) 'The 2012 Ageing Report-Economic and budgetary projections for the 27 EU Member States (2010-2060).' European Economy (2)
- Fletcher, J. M., and S. F. Lehrer (2009) 'Using genetic lotteries within families to examine the causal impact of poor health on academic achievement.' Technical Report, National Bureau of Economic Research
- Fredman, L., J. A. Cauley, S. Satterfield, E. Simonsick, S. M. Spencer, H. N. Ayonayon, T. B. Harris et al. (2008) 'Caregiving, mortality, and mobility decline: the Health, Aging, and Body Composition (Health ABC) study.' Archives of internal medicine 168(19), 2154
- Glei, D. A., D. A. Landau, N. Goldman, Y-L. Chuang, G. Rodríguez, and M. Weinstein (2005) 'Participating in social activities helps preserve cognitive function: an analysis of a longitudinal, population-based study of the elderly.' *International Journal of Epidemiology* 34(4), 864–871
- Glendinning, C., H. Arksey, F. Tjadens, M. Morée, N. Moran, and H. Nies (2009) 'Care provision within families and its socio-economic impact on care providers across the European Union.' *Research Works* No. 5
- Green, A. F., G. Rebok, and C. G. Lyketsos (2008) 'Influence of social network characteristics on cognition and functional status with aging.' *International journal of geriatric psychiatry* 23(9), 972–978
- Grossman, M. (1972) 'On the concept of health capital and the demand for health.' The Journal of Political Economy 80, 223–255
- Haberkern, K., and M. Szydlik (2010) 'State care provision, societal opinion and children's care of older parents in 11 European countries.' Ageing and Society 30(2), 299
- He, D., and P. McHenry (2013) 'Does labor force participation reduce informal caregiving?' Available at SSRN 2292700

- Heger, D. (2014) 'Employment and well-being of informal caregivers.' In: Three essays on Informal Care, Health, and Education, Ph.D. Thesis, Queen's University. http://hdl.handle.net/1974/12236
- Heitmueller, A. (2007) 'The chicken or the egg?: Endogeneity in labour market participation of informal carers in England.' Journal of Health Economics 26(3), 536–559
- Heitmueller, A., and K. Inglis (2007) 'The earnings of informal carers: Wage differentials and opportunity costs.' Journal of Health Economics 26(4), 821–841
- Hirst, M. (2005) 'Carer distress: a prospective, population-based study.' Social Science & Medicine 61(3), 697– 708
- Hsu, H. C. (2007) 'Does social participation by the elderly reduce mortality and cognitive impairment?' Aging & mental health 11(6), 699–707
- Idler, E. L., and Y. Benyamini (1997) 'Self-rated health and mortality: A review of twenty-seven community studies.' Journal of Health and Social Behavior 38(1), 21–37
- Jürges, H. (2007) 'True health vs response styles: Exploring cross-country differences in self-reported health.' Health Economics 16, 163–178
- Jürges, H., M. Avendano, and J. P. Mackenbach (2008) 'Are different measures of self-rated health comparable? An assessment in five European countries.' *European journal of epidemiology* 23(12), 773–781
- King, D., and L. Pickard (2013) 'When is a carer's employment at risk?: longitudinal analysis of unpaid care and employment in midlife in England.' *Health and social care in the community* 21(3), 303–314
- Lee, S., I. Kawachi, and F. Grodstein (2004) 'Does caregiving stress affect cognitive function in older women?' The Journal of nervous and mental disease 192(1), 51–57
- Leipold, B., C. Schacke, and S. Zank (2008) 'Personal growth and cognitive complexity in caregivers of patients with dementia.' European Journal of Ageing 5(3), 203–214
- Lilly, M. B., A. Laporte, and P. C. Coyte (2007) 'Labor market work and home care's unpaid caregivers: a systematic review of labor force participation rates, predictors of labor market withdrawal, and hours of work.' Milbank Quarterly 85(4), 641–690
- Mackenzie, C. S., U. J. Wiprzycka, L. Hasher, and D. Goldstein (2009) 'Associations between psychological distress, learning, and memory in spouse caregivers of older adults.' *The Journals of Gerontology Series* B: Psychological Sciences and Social Sciences 64(6), 742–746
- Maddox, G. L., and E. B. Douglass (1973) 'Self-assessment of health: A longitudinal study of elderly subjects.' Journal of Health and Social Behavior 14(1), 87–93
- Matire, L. M., and R. Schulz (2012) 'Informal caregiving to older adults: Health effects of providing and receiving care.' In 'Handbook of Health Psychology,' 2 ed. (Psychology Press) pp. 293–308
- Meng, A. (2013) 'Informal home care and labor-force participation of household members.' Empirical Economics 44(2), 959–979
- OECD (2005) Long-term care for older people (OECD, Paris)
- Pinquart, M., and S. Sörensen (2003) 'Associations of stressors and uplifts of caregiving with caregiver burden and depressive mood: a meta-analysis.' The Journals of Gerontology Series B: Psychological Sciences and Social Sciences 58(2), P112–P128
- (2005) 'Ethnic differences in stressors, resources, and psychological outcomes of family caregiving: A metaanalysis.' The Gerontologist 45(1), 90–106
- (2007) 'Correlates of physical health of informal caregivers: a meta-analysis.' The Journals of Gerontology Series B: Psychological Sciences and Social Sciences 62(2), P126–P137

- Schaffer, M. E. (2012) 'xtivreg2: Stata module to perform extended IV/2SLS, GMM and AC/HAC, LIML and k-class regression for panel data models.' *Statistical Software Components*
- Schulz, R., J. Newsom, M. Mittelmark, L. Burton, C. Hirsch, and S. Jackson (1997) 'Health effects of caregiving: The caregiver health effects study: An ancillary study of the cardiovascular health study.' Annals of Behavioral Medicine 19(2), 110–116
- Stock, J. H., and M. Yogo (2005) 'Testing for weak instruments in linear IV regression.' Identification and inference for econometric models: Essays in honor of Thomas Rothenberg
- Van Houtven, C. H., N. B. Coe, and M. Skira (2013) 'The effect of informal care on work and wages.' Journal of Health Economics 32(1), 240 – 252
- Viitanen, T. (2010) 'Informal eldercare across Europe: Estimates from the European Community Household Panel.' Economic Analysis and Policy 40(2), 149–178
- Vitaliano, P., D. Echeverria, M. Shelkey, J. Zhang, and J. Scanlan (2007) 'A cognitive psychophysiological model to predict functional decline in chronically stressed older adults.' *Journal of Clinical Psychology* in Medical Settings 14(3), 177–190
- Wooldridge, J. M. (2002) Econometric analysis of cross section and panel data (MIT press)
- Ziebarth, N. (2010) 'Measurement of health, the sensitivity of the concentration index, and reporting heterogeneity.' Social Science and Medicine 71, 116–124

#### Appendix

#### A Assessing reporting bias in parental health

Systematic misreporting of parental health would question the reliability of using the variable as an instrument, for example, if daughters misstate their parents' health to justify not providing informal care. As daughters living close to their parents may feel more pressured to justify not providing care to their parents, I can test for such behaviour by comparing parental health for parents living close to their daughters and parents living further away from their daughters. Results are presented in Table 7. I find no significant differences in parental health based on the distance to the daughter. Besides, the question about parental health is asked at the beginning of the SHARE interview, whereas the questions about caregiving behaviour are asked later on, which further limits the possibility for strategic answering behaviour.

Table 7: Parental health and distance to the parent

Distance to the parent

Parental health	$\leq 5 \text{ km}$	$>5~{\rm km}$		${\leq}25~{\rm km}$	$>\!25~{\rm km}$	
poor	0.29	0.31	(0.290)	0.29	0.31	(0.286)
fair	0.34	0.37	(0.395)	0.35	0.37	(0.474)
good	0.29	0.25	(0.132)	0.29	0.25	(0.135)
very good	0.07	0.07	(0.485)	0.07	0.07	(0.648)

Weighted values for individuals with at least on living parent based on individuals' first observation. Two sided p-values of adjusted Wald test of equality of means in parentheses. N=3,172

#### **B** Additional tables

Country	Retirement age	Comments
Austria	60	
Belgium	65	Workers can retire at age 60 with 40 years of con- tributions.
Czech Republic	61	
Denmark	65	
France	60	Workers can retire at age 60 with 40 years of con- tributions.
Germany	65	
Greece	62	Workers can retire at age 59 with 35 years of con- tributions.
Italy	60	Workers can retire at 57 (56 for manual workers) with 35 years of contributions.
Netherlands	65	
Poland	60	
Spain	65	
Switzerland	64	
Sweden	65	

#### Table 8: Official retirement age for women in SHARE countries

Source: OECD Statistics on average effective age and official age of retirement in OECD countries (official retirement age for 2010)

	Number of individuals
Never caregiver	1,653
Ever caregiver	1,656
Continuous caregivers	496
Begin caregiving	504
End caregiving	795
Ever weekly caregiver	1,164
Continuous caregivers	294
Begin caregiving	417
End caregiving	551
Ever daily caregiver	550
Continuous caregivers	118
Begin caregiving	215
End caregiving	263

Table 9: Changes in caregiving behaviour over time

		Employed			Η	lours work	ed								
Frequency of care:	any	weekly	daily	an	ıy	weekly	daily								
Caregiver	-0.028	-0.013	-0.016	2.0	32	0.911	1.714								
(family care country)	(0.025)	(0.027)	(0.030)	(1.4	82)	(1.506)	(1.602)								
Caregiver	0.042**	$0.042^{*}$	-0.021	0.4	41	0.054	0.082								
(formal care country)	(0.020)	(0.024)	(0.037)	(0.8)	45)	(1.095)	(2.266)								
Chronic conditions	-0.004	-0.005	-0.005	-0.9	06*	-0.916*	-0.928*								
	(0.007)	(0.007)	(0.007)	(0.4)	94)	(0.501)	(0.505)								
ADL	-0.006	-0.006	-0.005	-1.1	33	-1.125	-1.098								
	(0.019)	(0.019)	(0.019)	(1.0	14)	(1.009)	(0.999)								
IADL	0.024	0.023	0.023	1.6	$45^{-1}$	1.671	1.668								
	(0.017)	(0.017)	(0.017)	(1.3)	21)	(1.333)	(1.332)								
Age	-0.113	-0.112	-0.107	4.1	85	4.091	4.203								
	(0.074)	(0.075)	(0.075)	(7.9)	(90)	(7.937)	(7.941)								
Age squared/100	0.114*	0.114*	$0.109^{*}$	-4.0	)57	-4.024	-4.085								
	(0.062)	(0.062)	(0.062)	(6.7)	91)	(6.811)	(6.819)								
Wave 2	-0.104***	-0.107***	-0.105***	-0.0	)33	0.050	-0.017								
	(0.035)	(0.035)	(0.035)	(1.9)	24)	(1.939)	(1.948)								
Wave 4	-0.219**	-0.228**	-0.223**	2.4	85	2.800	2.525								
	(0.097)	(0.097)	(0.097)	(5.0)	27)	(5.051)	(5.072)								
Married	0.031	0.031	0.030	-2.3	867	-2.318	-2.321								
	(0.049)	(0.049)	(0.049)	(1.5)	13)	(1.488)	(1.485)								
Household size	-0.026**	-0.026**	-0.026**	0.4	99	0.556	0.556								
	(0.011) 0.019	(0.011) 0.019	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.6)	(01)	(0.596)	(0.597)
1st wealth quartile			0.018	0.018	-0.4	0.431	-0.496	-0.479							
	(0.030)	(0.030)	(0.031)	(1.5)	(05)	(1.493)	(1.494)								
2nd wealth quartile	0.017	0.017	0.017	-1.1	59	-1.176	-1.174								
	(0.024)	(0.024)	(0.024)	(1.2)	53)	(1.249)	(1.255)								
3rd wealth quartile	0.015	0.015	0.015	-1.3	347	-1.342	-1.320								
	(0.021)	(0.021)	(0.021)	(1.0)	(05)	(0.999)	(1.002)								
Reached off. ret. age	-0.076***	-0.075***	-0.077***	-2.1	37	-2.323	-2.173								
	(0.028)	(0.028)	(0.028)	(2.8)	94)	(2.916)	(2.882)								
2 years to off. ret. age	-0.067***	-0.068***	-0.068***	-1.0	)51	-1.009	-1.044								
	(0.025)	(0.025)	(0.025)	(1.5)	42)	(1.532)	(1.548)								
5 years to off. ret. age	-0.076***	-0.075***	$-0.074^{***}$	-1.5	545	-1.518	-1.503								
	(0.026)	(0.026)	(0.026)	(1.2)	34)	(1.231)	(1.234)								
Observations	7,559	7,559	7,559	$^{3,1}$	43	3,143	3,143								
Within R-squared	0.110	0.109	0.108	0.0	66	0.063	0.063								
Unique individuals	3,296	3,296	3,296	1,4	09	1,409	1,409								

Table 10: FE results: Labour force participation

Standard errors clustered at household level in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Regressions include age\*country and age squared/100\*country interaction terms.

τ	daily	0.324	(0.364)	-0.302	(0.445)			$-0.465^{***}$	(0.127)	-0.484	(0.381)	-0.162	(0.270)	$1.967^{*}$	(1.060)	$-1.442^{*}$	(0.868)	-0.510	(0.527)	$-2.492^{*}$	(1.470)	-0.154	(0.738)	0.133	(0.170)	-0.166	(0.365)	-0.251	(0.256)	-0.137	(0.379)	-0.155	next page
Grip strengtl	weekly	$0.667^{**}$	(0.335)	-0.499*	(0.289)			$-0.471^{***}$	(0.127)	-0.475	(0.383)	-0.164	(0.270)	$2.001^{*}$	(1.061)	$-1.490^{*}$	(0.870)	-0.482	(0.527)	-2.389	(1.470)	-0.189	(0.733)	0.126	(0.169)	-0.149	(0.365)	-0.264	(0.255)	-0.175	(0.378)	-0.154	Continued or
0	any	$0.646^{**}$	(0.306)	-0.343	(0.243)			$-0.469^{***}$	(0.127)	-0.464	(0.383)	-0.173	(0.270)	$2.004^{*}$	(1.063)	-1.473*	(0.871)	-0.530	(0.528)	-2.542*	(1.472)	-0.184	(0.734)	0.119	(0.170)	-0.137	(0.364)	-0.267	(0.256)	-0.155	(0.377)	-0.149	
alth	daily	0.069	(0.065)	-0.133	(0.086)			$0.161^{***}$	(0.018)	0.046	(0.041)	$0.116^{***}$	(0.036)	-0.178	(0.218)	0.135	(0.185)	$0.219^{**}$	(0.095)	$0.468^{*}$	(0.255)	-0.018	(0.110)	0.003	(0.028)	0.001	(0.045)	$-0.094^{**}$	(0.037)	0.016	(0.063)	0.003	
perceived he	weekly	0.035	(0.058)	$-0.111^{**}$	(0.051)			$0.161^{***}$	(0.018)	0.047	(0.040)	$0.114^{***}$	(0.036)	-0.171	(0.219)	0.127	(0.185)	$0.223^{**}$	(0.095)	0.477*	(0.256)	-0.023	(0.1111)	0.001	(0.028)	0.003	(0.045)	-0.094**	(0.037)	0.017	(0.063)	0.004	
Self-	any	0.002	(0.053)	$-0.109^{**}$	(0.043)			$0.161^{***}$	(0.018)	0.048	(0.040)	$0.113^{***}$	(0.036)	-0.171	(0.220)	0.128	(0.186)	$0.217^{**}$	(0.095)	$0.459^{*}$	(0.256)	-0.021	(0.111)	0.001	(0.028)	0.004	(0.045)	-0.093**	(0.037)	0.019	(0.063)	0.005	
	daily	0.115	(0.174)	0.328	(0.313)	0.064	(0.106)	$0.227^{***}$	(0.055)	-0.070	(0.132)	$0.332^{**}$	(0.151)	-0.433	(0.444)	0.334	(0.357)	-0.111	(0.255)	-0.007	(0.693)	-0.577	(0.366)	0.007	(0.096)	0.066	(0.134)	-0.110	(0.101)	0.182	(0.176)	-0.047	
EURO-D	weekly	0.204	(0.150)	-0.014	(0.126)	0.071	(0.106)	$0.225^{***}$	(0.055)	-0.071	(0.132)	$0.333^{**}$	(0.151)	-0.422	(0.446)	0.320	(0.359)	-0.106	(0.256)	0.024	(0.694)	-0.573	(0.367)	0.010	(0.097)	0.066	(0.134)	-0.113	(0.101)	0.171	(0.175)	-0.045	
	any	0.202	(0.133)	0.104	(0.110)	0.077	(0.107)	$0.226^{***}$	(0.055)	-0.069	(0.132)	$0.331^{**}$	(0.151)	-0.421	(0.445)	0.325	(0.358)	-0.111	(0.256)	0.004	(0.695)	-0.569	(0.366)	0.008	(0.097)	0.066	(0.134)	-0.117	(0.101)	0.176	(0.176)	-0.047	
	Frequency of care:	Caregiver	(family care country)	Caregiver	(formal care country)	Parental loss		Chronic conditions		ADL		IADL		Age		Age squared/ $100$		Wave 2		Wave 4		Married		Household size		Employed		Other activities		1st wealth quartile		2nd wealth quartile	

Table 11: FE results: Health

Table 11 – continued from previous page	EURO-D Self-perceived health Grip strength	of care: any weekly daily any weekly daily any weekly daily	(0.145) $(0.145)$ $(0.145)$ $(0.145)$ $(0.050)$ $(0.050)$ $(0.050)$ $(0.306)$ $(0.307)$ $(0.307)$	quartile -0.181 -0.181 -0.179 -0.017 -0.018 -0.020 -0.240 -0.248 -0.245	(0.124) $(0.123)$ $(0.123)$ $(0.046)$ $(0.046)$ $(0.045)$ $(0.261)$ $(0.262)$ $(0.261)$	us 7,274 7,274 7,274 7,526 7,526 7,526 7,518 7,018 7,018	quared $0.037$ $0.037$ $0.037$ $0.037$ $0.080$ $0.080$ $0.080$ $0.058$ $0.058$	viduals 3,180 3,180 3,180 3,282 3,282 3,282 3,075 3,075 3,075	Tors clustered at household level in parentheses. *** $p<0.01$ , ** $p<0.05$ , * $p<0.1$	include age*country and age squared/100*country interaction terms.
		Frequency of care:		3rd wealth quartile		Observations	Within R-squared	Unique individuals	Standard errors clustere	Regressions include age <sup>*</sup>

	Ve	rbal fluenc	у	Short-	term word	recall
Frequency of care:	any	weekly	daily	any	weekly	daily
Caregiver	1.136***	0.845**	1.158**	0.042	-0.050	-0.145
(family care country)	(0.389)	(0.425)	(0.504)	(0.111)	(0.126)	(0.132)
Caregiver	-0.235	0.329	-0.503	0.093	0.214**	0.283**
(formal care country)	(0.339)	(0.421)	(0.771)	(0.088)	(0.093)	(0.134)
Chronic conditions	$0.238^{*}$	$0.238^{*}$	$0.240^{*}$	0.010	0.011	0.012
	(0.133)	(0.134)	(0.134)	(0.039)	(0.039)	(0.039)
ADL	-0.177	-0.202	-0.197	0.003	0.001	0.004
	(0.251)	(0.251)	(0.253)	(0.083)	(0.084)	(0.083)
IADL	$-0.419^{*}$	-0.406*	$-0.387^{*}$	0.021	0.021	0.017
	(0.217)	(0.218)	(0.218)	(0.066)	(0.066)	(0.067)
Age	1.515	1.432	1.504	0.349	0.334	0.346
	(1.184)	(1.172)	(1.178)	(0.320)	(0.320)	(0.318)
Age squared/100	-1.482	-1.418	-1.477	-0.249	-0.233	-0.247
	(0.966)	(0.956)	(0.962)	(0.265)	(0.266)	(0.264)
Wave 2	1.085	1.117	1.106	0.076	0.069	0.076
	(0.737)	(0.737)	(0.739)	(0.178)	(0.177)	(0.178)
Wave 4	1.478	1.588	1.544	0.067	0.043	0.059
	(1.979)	(1.979)	(1.983)	(0.483)	(0.480)	(0.482)
Married	-0.362	-0.311	-0.349	$0.641^{**}$	$0.652^{**}$	$0.642^{**}$
	(0.770)	(0.765)	(0.762)	(0.253)	(0.253)	(0.253)
Household size	-0.084	-0.063	-0.059	-0.054	-0.053	-0.057
	(0.172)	(0.174)	(0.173)	(0.050)	(0.051)	(0.051)
Employed	0.074	0.037	0.045	0.100	0.096	0.102
	(0.350)	(0.351)	(0.350)	(0.092)	(0.092)	(0.091)
Other activities	0.168	0.189	0.192	-0.062	-0.058	-0.058
	(0.305)	(0.305)	(0.305)	(0.079)	(0.079)	(0.080)
1st wealth quartile	-0.232	-0.226	-0.227	0.072	0.080	0.084
	(0.446)	(0.449)	(0.442)	(0.122)	(0.121)	(0.122)
2nd wealth quartile	-0.134	-0.127	-0.144	0.114	0.116	0.117
	(0.333)	(0.335)	(0.333)	(0.111)	(0.111)	(0.111)
3rd wealth quartile	-0.065	-0.061	-0.080	-0.079	-0.077	-0.074
	(0.307)	(0.308)	(0.307)	(0.087)	(0.086)	(0.086)
Observations	7 501	7 501	7 501	7 516	7 516	7 516
Within R-squared	0.040	0.038	0.038	0.047	0.048	0.048
Unique individuals	3 270	3.270	3.000	3 277	3.277	3 277
	0,210	0,210	0,210	 3,211	0,211	0,211

Table 12:	FE results:	Cognitive	ability	(I)

Standard errors clustered at household level in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1. Regressions include age\*country and age squared/100\*country interaction terms.

	Long-	term word	l recall		Numeracy	τ
Frequency of care:	any	weekly	daily	 any	weekly	daily
Caregiver	-0.132	-0.232*	-0.264*	0.014	0.034	-0.070
(family care country)	(0.109)	(0.125)	(0.136)	(0.044)	(0.048)	(0.062)
Caregiver	0.120	$0.168^{*}$	$0.331^{*}$	-0.012	-0.029	-0.014
(formal care country)	(0.090)	(0.092)	(0.189)	(0.033)	(0.035)	(0.055)
Chronic conditions	-0.026	-0.024	-0.024	-0.020	-0.020	-0.019
	(0.043)	(0.043)	(0.043)	(0.014)	(0.014)	(0.014)
ADL	0.026	0.027	0.030	0.010	0.010	0.008
	(0.087)	(0.088)	(0.087)	(0.035)	(0.035)	(0.035)
IADL	-0.026	-0.028	-0.034	-0.021	-0.021	-0.021
	(0.083)	(0.084)	(0.083)	(0.040)	(0.040)	(0.040)
Age	0.023	0.020	0.023	0.233	0.235	0.232
	(0.423)	(0.422)	(0.426)	(0.155)	(0.155)	(0.154)
Age squared/100	0.170	0.179	0.171	-0.218*	-0.220*	-0.216*
	(0.352)	(0.351)	(0.355)	(0.125)	(0.125)	(0.124)
Wave 2	-0.118	-0.131	-0.122	0.135	0.137	0.137
	(0.202)	(0.202)	(0.201)	(0.095)	(0.095)	(0.095)
Wave 4	-0.439	-0.484	-0.457	0.226	0.232	0.228
	(0.552)	(0.552)	(0.551)	(0.245)	(0.244)	(0.244)
Married	0.454	0.458	0.451	0.095	0.094	0.099
	(0.415)	(0.415)	(0.415)	(0.107)	(0.107)	(0.107)
Household size	-0.047	-0.049	-0.053	-0.005	-0.005	-0.005
	(0.061)	(0.061)	(0.060)	(0.021)	(0.021)	(0.021)
Employed	-0.142	-0.141	-0.136	0.060	0.060	0.058
	(0.112)	(0.113)	(0.112)	(0.045)	(0.045)	(0.045)
Other activities	0.036	0.037	0.035	-0.003	-0.004	-0.002
	(0.088)	(0.088)	(0.088)	(0.029)	(0.029)	(0.029)
1st wealth quartile	-0.014	-0.003	-0.007	0.005	0.003	0.010
	(0.141)	(0.140)	(0.140)	(0.056)	(0.055)	(0.056)
2nd wealth quartile	0.010	0.011	0.015	0.016	0.016	0.017
	(0.120)	(0.120)	(0.120)	(0.044)	(0.044)	(0.044)
3rd wealth quartile	$-0.199^{*}$	$-0.197^{*}$	$-0.193^{*}$	0.005	0.005	0.006
	(0.104)	(0.104)	(0.104)	(0.040)	(0.040)	(0.040)
Observations	7,516	7,516	7,516	7,514	7,514	7,514
Within R-squared	0.077	0.078	0.079	0.017	0.018	0.018
Unique individuals	3,277	3,277	3,277	3,277	3,277	3,277

Table 13: FE results: Cognitive ability (II)

Standard errors clustered at household level in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1. Regressions include age\*country and age squared/100\*country interaction terms.

$ \begin{array}{ccccc} -0.344^{**} & -0.340^{**} & -0.370^{**} & -0.370^{**} & -0.600^{**} & -0.600^{**} \\ \text{is country} & (0.114) & (0.14) & (0.168) & (0.168) & (0.237) & (0.237) \\ 0.001 & 0.001 & 0.013 & 0.003 & (0.003) & (0.003) & (0.003) & (0.003) \\ -0.011 & -0.013 & 0.013 & 0.003 & (0.003) & (0.003) & (0.003) & (0.003) \\ 0.0120 & (0.012) & (0.013) & (0.013) & (0.001) & (0.003) & (0.003) & (0.003) \\ 0.0120 & (0.013) & (0.013) & (0.013) & (0.011) & (0.013) & (0.003) & (0.003) \\ 0.0120 & (0.013) & (0.013) & (0.013) & (0.011) & (0.013) & (0.003) & (0.003) \\ 0.0120 & (0.013) & (0.013) & (0.013) & (0.011) & (0.013) & (0.003) & (0.011) & (0.013) & (0.003) \\ 0.0130 & (0.013) & (0.013) & (0.013) & (0.013) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.013) & (0.003) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.013) & (0.012) & (0.0$	ndent variable:	Employed	$family^a$	$formal^b$	Employed	family	formal	Employed	family	formal
v country) $(0.144)$ $(0.143)$ $(0.143)$ $(0.123)$ $(0.227)$ anditions $(0.001)$ $(0.003)$ $(0.$		-0.344**			$-0.370^{**}$			-0.600**		
te country) (0.149) (0.194) (0.194) (0.134) (0.257) (0.257) (0.008) (0.003) (0.005) (0.003) (0.013) (0.003) (0.013) (0.003) (0.013) (0.003) (0.013) (0.003) (0.013) (0.003) (0.013) (0.013) (0.003) (0.017) (0.011) (0.021) (0.013) (0.007) (0.003) (0.007) (0.003) (0.013) (0.003) (0.013) (0.003) (0.017) (0.013) (0.003) (0.017) (0.013) (0.003) (0.017) (0.013) (0.003) (0.017) (0.013) (0.003) (0.017) (0.013) (0.003) (0.017) (0.013) (0.003) (0.017) (0.013) (0.003) (0.017) (0.013) (0.003) (0.017) (0.003) (0.012) (0.003) (0.013) (0.003) (0.017) (0.003) (0.013) (0.013) (0.003) (0.012) (0.013) (	e country)	(0.144) 0.018			(0.168) 0.023			(0.272) 0.038		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	re country)	(0.149)			(0.194)			(0.257)		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	onditions	0.001	0.013	-0.001	0.004	$0.018^{**}$	0.002	0.003	0.010	-0.001
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.008)	(0.009)	(0.005)	(0.008)	(0.008)	(0.004)	(0.009)	(0.006)	(0.003)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		-0.011	-0.018	0.018	-0.006	-0.005	0.013	-0.007	-0.005	-0.004
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.020)	(0.018)	(0.013)	(0.020)	(0.017)	(0.011)	(0.021)	(0.013)	(0.007)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		0.028	0.019	-0.006	0.024	0.007	0.001	0.016	-0.008	0.007
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.018)	(0.015)	(0.000)	(0.017)	(0.014)	(0.007)	(0.018)	(0.010)	(0.005)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		$-0.153^{*}$	-0.005	0.068	$-0.152^{*}$	-0.002	0.050	$-0.150^{*}$	0.002	0.023
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.079)	(0.036)	(0.127)	(0.079)	(0.033)	(0.107)	(0.079)	(0.027)	(0.070)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	ed/100	$0.150^{**}$	0.012	-0.074	$0.158^{**}$	0.030	-0.065	$0.148^{**}$	0.004	-0.025
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.065)	(0.028)	(0.105)	(0.067)	(0.026)	(0.091)	(0.066)	(0.023)	(0.066)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		$-0.110^{***}$	0.005	-0.016	$-0.126^{***}$	-0.036	0.034	$-0.112^{***}$	0.000	0.011
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.038)	(0.029)	(0.044)	(0.039)	(0.026)	(0.035)	(0.038)	(0.020)	(0.033)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		$-0.245^{**}$	-0.039	-0.066	$-0.302^{***}$	$-0.182^{**}$	0.083	$-0.254^{**}$	-0.034	0.038
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.105)	(0.093)	(0.107)	(0.111)	(0.085)	(0.087)	(0.108)	(0.071)	(0.080)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		0.030	0.068	-0.013	0.039	$0.086^{**}$	-0.030	0.039	0.056	0.018
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.053)	(0.044)	(0.049)	(0.055)	(0.040)	(0.048)	(0.056)	(0.037)	(0.025)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	l size	$-0.028^{**}$	0.016	-0.004	$-0.035^{***}$	-0.005	-0.005	$-0.034^{**}$	0.000	0.012
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.013)	(0.014)	(0.008)	(0.013)	(0.013)	(0.007)	(0.014)	(0.010)	(100.0)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	ı quartile	0.042	$0.072^{**}$	-0.002	0.055	$0.101^{***}$	-0.006	0.055	$0.062^{**}$	-0.003
h quartile $0.021$ $0.004$ $0.015$ $0.025$ $0.015$ $-0.001$ $0.031$ $0.019$ $0.000$ a quartile $0.027$ $(0.024)$ $(0.017)$ $(0.027)$ $(0.024)$ $(0.014)$ $(0.028)$ $(0.020)$ $(0.011)$ a quartile $0.016$ $0.007$ $0.010$ $0.019$ $0.013$ $-0.000$ $0.025$ $0.017$ $-0.006$ bff. ret. age $-0.055^{***}$ $0.012$ $0.011$ $-0.087^{***}$ $0.006$ $-0.036^{**}$ $-0.030^{**}$ $(0.023)$ off. ret. age $-0.067^{***}$ $0.011$ $-0.087^{***}$ $0.006$ $-0.036^{**}$ $-0.076$ $(0.023)$ $(0.023)$ off. ret. age $-0.067^{***}$ $-0.011$ $-0.087^{***}$ $0.006$ $-0.036^{**}$ $-0.076$ $(0.023)$ off. ret. age $-0.067^{***}$ $-0.014$ $-0.009$ $-0.088^{***}$ $-0.018$ $0.010$ $(0.020)$ $(0.026)$ $(0.023)$ off. ret. age $-0.067^{***}$ $-0.011$ $-0.028$ $(0.028)$ $(0.022)$ $(0.018)$ $(0.029)$ $(0.018)$ $(0.023)$ off. ret. age $-0.078^{****}$ $-0.009$ $-0.048^{****}$ $-0.025$ $0.011$ $-0.079^{****}$ $-0.007$ $-0.007$ 0.013 $(0.023)$ $(0.021)$ $(0.028)$ $(0.022)$ $(0.011)$ $(0.029)$ $(0.018)$ $(0.017)off. ret. age -0.078^{****} -0.009 0.045^{****} -0.088^{****} -0.025 0.011 -0.079^{****} -0.007 0.013^{*}$		(0.036)	(0.031)	(0.021)	(0.038)	(0.028)	(0.016)	(0.039)	(0.025)	(0.013)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	h quartile	0.021	0.004	0.015	0.025	0.015	-0.001	0.031	0.019	0.000
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.027)	(0.024)	(0.017)	(0.027)	(0.024)	(0.014)	(0.028)	(0.020)	(0.011)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	h quartile	0.016	0.007	0.010	0.019	0.013	-0.000	0.025	0.017	-0.006
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.023)	(0.023)	(0.016)	(0.023)	(0.021)	(0.011)	(0.025)	(0.018)	(0.008)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	off. ret. age	-0.085***	0.012	-0.011	-0.087***	0.006	$-0.036^{*}$	-0.080**	0.016	0.002
off. ret. age $-0.066^{**}$ $-0.014$ $-0.009$ $-0.068^{**}$ $-0.018$ $0.010$ $-0.065^{**}$ $-0.007$ $-0.006$ $0.010$ $0.028)$ $(0.028)$ $(0.020)$ $(0.018)$ $(0.017)$ $(0.017)$ $(0.028)$ $(0.018)$ $(0.017)$ $(0.017)$ $(0.078^{***}$ $-0.009$ $0.045^{***}$ $-0.084^{***}$ $-0.025$ $0.011$ $-0.079^{***}$ $-0.007$ $0.013^{**}$ $(0.028)$ $(0.011)$ $(0.029)$ $(0.021)$ $(0.013)$ $(0.012)$ $(0.028)$ $(0.028)$ $(0.011)$ $(0.029)$ $(0.021)$ $(0.021)$ $(0.007)$ $(0.013)$		(0.032)	(0.031)	(0.023)	(0.032)	(0.030)	(0.020)	(0.034)	(0.026)	(0.023)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	off. ret. age	$-0.066^{**}$	-0.014	-0.009	-0.068**	-0.018	0.010	-0.065**	-0.007	-0.006
off. ret. age $-0.078^{***}$ $-0.009$ $0.045^{***}$ $-0.084^{***}$ $-0.025$ $0.011$ $-0.079^{***}$ $-0.007$ $0.013^{*}$ $(0.028)$ $(0.028)$ $(0.021)$ $(0.021)$ $(0.021)$ $(0.007)$		(0.028)	(0.024)	(0.021)	(0.028)	(0.022)	(0.018)	(0.029)	(0.018)	(0.017)
(0.028) $(0.027)$ $(0.015)$ $(0.028)$ $(0.025)$ $(0.011)$ $(0.029)$ $(0.021)$ $(0.007)$	off. ret. age	-0.078***	-0.009	$0.045^{***}$	-0.084***	-0.025	0.011	-0.079***	-0.007	$0.013^{*}$
		(0.028)	(0.027)	(0.015)	(0.028)	(0.025)	(0.011)	(0.029)	(0.021)	(0.007)

		Tab	le 14 – cont	tinued from p	previous pag	je.			
Dependent variable:	Employed	family	formal	Employed	family	formal	Employed	family	formal
Parent in poor health		$0.198^{***}$	0.003		$0.184^{***}$	0.000		$0.113^{***}$	-0.001
(family care country)		(0.034)	(0.002)		(0.034)	(0.002)		(0.026)	(0.001)
Parent in poor health		0.001	$0.153^{***}$		0.001	$0.117^{***}$		0.002	$0.089^{***}$
(formal care country)		(0.003)	(0.034)		(0.003)	(0.027)		(0.002)	(0.019)
Observations	6,991	6,991	6,991	6,991	6,991	6,991	6,991	6,991	6,991
Within R-squared	0.030	0.096	0.055	0.018	0.083	0.034	-0.058	0.050	0.028
Unique individuals	3,070	3,070	3,070	3,070	3,070	3,070	3,070	3,070	3,070
Standard errors cluster	red at househe	old level in <sub>1</sub>	parentheses	. *** p<0.01	, ** p<0.05	6, * p<0.1			
Regressions include age	e*country and	age sonare	d /100*cour	ntry interaction	on terms.				

Table 14 – continued from previous page

<sup>a</sup> Caregiver in family care country and age squared too country interact

formal			-0.002	(0.007)	-0.003	(0.009)	0.017	(0.014)	-0.080	(0.161)	0.075	(0.144)	-0.003	(0.032)	-0.005	(0.076)	0.010	(0.045)	$0.011^{**}$	(0.005)	-0.013	(0.021)	0.014	(0.017)	-0.004	(0.012)	-0.049	(0.048)	-0.011	(0.041)	0.007	(0.017)	
family			0.017	(0.011)	-0.006	(0.015)	0.000	(0.015)	-0.021	(0.065)	0.019	(0.064)	-0.008	(0.029)	0.003	(0.104)	0.074	(0.068)	0.000	(0.013)	-0.011	(0.020)	-0.004	(0.022)	-0.023	(0.023)	$-0.112^{*}$	(0.066)	0.036	(0.033)	-0.013	(0.029)	
Hours	2.054 (11.609)	5.354	(14.459) -0.773	(0.584)	-0.898	(1.293)	1.484	(1.479)	-0.773	(6.002)	-0.074	(5.171)	0.604	(2.014)	3.765	(5.295)	$-3.245^{*}$	(1.707)	0.338	(0.646)	-1.263	(1.590)	-1.720	(1.364)	-1.257	(1.116)	-2.190	(3.503)	-0.835	(1.723)	-1.442	(1.318)	Č
formal			0.003	(0.010)	0.007	(0.022)	0.007	(0.013)	0.172	(0.280)	-0.162	(0.251)	-0.006	(0.047)	-0.023	(0.116)	0.043	(0.078)	0.001	(0.010)	-0.037	(0.032)	-0.013	(0.024)	-0.014	(0.019)	-0.061	(0.062)	-0.029	(0.052)	-0.011	(0.024)	
family			0.014	(0.015)	0.017	(0.027)	-0.010	(0.028)	0.042	(0.078)	-0.005	(0.073)	-0.061	(0.039)	-0.222*	(0.122)	$0.157^{**}$	(0.078)	-0.008	(0.019)	0.046	(0.035)	0.014	(0.037)	-0.006	(0.031)	-0.042	(0.079)	0.039	(0.049)	-0.001	(0.040)	
Hours	1.928 (10.637)	5.585 (17 707)	(15.792)	(0.559)	-1.002	(1.284)	1.555	(1.461)	-2.288	(6.155)	1.279	(5.463)	0.722	(2.132)	4.300	(5.763)	-3.581	(2.301)	0.409	(0.650)	-1.238	(1.715)	-1.607	(1.302)	-1.238	(1.085)	-2.256	(3.339)	-0.735	(1.714)	-1.365	(1.280)	
$formal^b$			0.007	(0.011)	-0.018	(0.027)	0.008	(0.018)	-0.041	(0.327)	0.061	(0.292)	-0.090	(0.057)	-0.285**	(0.138)	0.052	(0.085)	-0.003	(0.011)	-0.019	(0.039)	0.024	(0.026)	-0.012	(0.020)	-0.044	(0.070)	-0.080	(0.059)	0.017	(0.030)	
$family^a$			-0.001	(0.017)	0.003	(0.038)	0.010	(0.036)	-0.040	(0.088)	0.017	(0.080)	0.042	(0.046)	0.140	(0.147)	0.088	(0.074)	0.025	(0.025)	-0.011	(0.049)	-0.015	(0.036)	-0.000	(0.032)	-0.103	(0.067)	0.055	(0.054)	-0.001	(0.043)	
Hours	2.474 (15.060)	2.480	(0.765 - 0.765)	(0.518)	-0.893	(1.316)	1.533	(1.411)	-1.043	(5.628)	0.172	(4.839)	0.690	(2.078)	4.105	(5.738)	-3.384*	(2.029)	0.343	(0.709)	-1.283	(1.587)	-1.677	(1.372)	-1.294	(1.078)	-2.316	(3.449)	-0.758	(1.885)	-1.472	(1.303)	
Dependent variable:	Caregiver (family care country)	Caregiver	(Iormal care country) Chronic conditions		ADL		IADL		Age		Age squared/ $100$		Wave 2		Wave 4		Married		Household size		1st wealth quartile		2nd wealth quartile		3rd wealth quartile		Reached off. ret. age		2 years to off. ret. age		5 years to off. ret. age		

Hours
results:
stage
second
and
First
15:
Table

Table 13 – continued from brevious base	Hours family formal Hours family formal Hours family formal	$0.102  0.013^{**}  0.144^{**}  0.001  0.131^{***}  0.003$	(0.071) $(0.006)$ $(0.068)$ $(0.004)$ $(0.049)$ $(0.003)$	$0.001  0.149^{***}  0.005  0.065^{*}  -0.002  0.070^{***}$	(0.007) $(0.044)$ $(0.007)$ $(0.039)$ $(0.005)$ $(0.027)$	2,916 $2,916$ $2,916$ $2,916$ $2,916$ $2,916$ $2,916$ $2,916$ $2,916$ $2,916$	0.066 0.115 0.041 0.053 0.088 0.022 0.061 0.098 0.051	1,313  1,313  1,313  1,313  1,313  1,313  1,313  1,313  1,313  1,313	red at household level in parentheses. $^{***}$ p<0.01, $^{**}$ p<0.05, $^{*}$ p<0.1	$e^*$ country and age squared/100 $^*$ country interaction terms.	and countrary b Conversion in formed concernations
Table to _ C	family form	0.102 $0.013$	(0.071) $(0.00)$	0.001 $0.149$	(0.04) (0.04)	2,916 2,91	0.115 $0.04$	1,313 $1,31$	shold level in pare	nd age squared/1	b Carearizer in f
	Hours					2,916	0.066	1,313	ered at house	ge*country a	minina anaa
	Dependent variable:	Parent in poor health	(family care country)	Parent in poor health	(formal care country)	Observations	Within R-squared	Unique individuals	Standard errors cluste	Regressions include ag	a Caroainar in family.

Table 15 – continued from previous page

40

amily formal	$\begin{array}{ccc} 0.025 & 0.010 \\ 0.017) & (0.012) \end{array}$	0.010 $-0.001$ $0.003$	0.001 $-0.0030.013$ $(0.007)$	0.008 0.006	$\begin{array}{rcc} 0.011) & (0.006) \\ 0.001 & 0.013 \end{array}$	(0.013) $(0.077)$	0.006 -0.016 -0.016 0.007)	0.003 0.007	0.020) (0.034)	0.038 $0.033$	0.073) (0.082)	0.062 $0.021$	0.039) (0.027)	0.010 $(0.008)$ $(0.008)$	0.006 -0.002	0.017) (0.010)	0.011 -0.001	0.013) (0.010)	.065** -0.003	0.026) (0.013)	0.021 -0.001	0.021) (0.011)	0.020 -0.006	0.019) (0.008)	and on nort news
EURO-D f 2.931* (1.610) 3.783** (1.547)	(0.122)	0.210***	-0.038 $-0.038$ $-$	0.305*	(0.157) ( -0.250 -	(0.535) (	0.142 (0.433) (	-0.065	(0.289) (	0.168	(0.793) (	-0.778*	(0.445) (0.003	(0.100)	0.103	(0.154) (	-0.107	(0.119) ()	-0.063 0	(0.238) (	-0.161	(0.174) (	-0.193	(0.146) ()	Conti
formal	0.000 (0.014)	(0.005)	0.014	-0.002	(0.008) 0.034	(0.106)	-0.054 (0.089)	0.042	(0.037)	0.098	(0.089)	-0.053	(0.045) 0.004	(0.007)	$0.025^{*}$	(0.014)	-0.001	(0.012)	-0.011	(0.017)	-0.003	(0.014)	-0.001	(0.011)	
family	-0.006 (0.018)	(0.008)	-0.003 (0.017)	0.009	(0.015) 0.007	(0.019)	$0.020^{*}$ (0.011)	-0.031	(0.026)	$-0.176^{**}$	(0.086)	$0.090^{**}$	(0.042)	(0.013)	0.011	(0.024)	0.017	(0.018)	$0.100^{***}$	(0.029)	0.011	(0.023)	0.012	(0.022)	
EURO-D 1.935* (1.022) 3.179**	$0.217^{*}$ 0.124	(0.063)	(0.153)	$0.292^{*}$	(0.162) -0.324	(0.572)	0.233 (0.466)	-0.120	(0.299)	0.209	(0.813)	-0.526	(0.472)	(0.104)	0.010	(0.164)	-0.108	(0.121)	-0.043	(0.230)	-0.115	(0.165)	-0.178	(0.138)	
$\mathbf{formal}^{b}$	-0.023 (0.015)	-0.002 (0.005)	0.021	-0.009	(0.010) 0.037	(0.125)	-0.046 (0.103)	-0.007	(0.045)	-0.055	(0.109)	-0.035	0.047)	(0.008)	$0.030^{*}$	(0.015)	0.008	(0.015)	-0.007	(0.022)	0.016	(0.017)	0.008	(0.017)	
family <sup>a</sup>	-0.007 (0.019)	(0.009)	-0.017 (0.018)	0.022	(0.015) -0.005	(0.022)	0.012 (0.012)	0.003	(0.030)	-0.046	(0.095)	0.073	(0.047)	(0.014)	-0.011	(0.025)	0.031	(0.021)	$0.074^{**}$	(0.032)	0.008	(0.024)	0.009	(0.023)	
EURO-D 1.849** (0.935) 2.585** (1.070)	$0.278^{**}$ (0.130)	$0.221^{***}$ (0.062)	-0.076	$0.287^{*}$	(0.160) -0.287	(0.549)	0.197 (0.442)	-0.033	(0.298)	0.405	(0.802)	-0.564	(0.435)	(0.102)	0.055	(0.163)	-0.158	(0.125)	-0.001	(0.220)	-0.156	(0.168)	-0.197	(0.145)	
Dependent variable: Caregiver (family care country) Caregiver (fermol concountry)	Parental loss	Chronic conditions	ADL	IADL	Age	)	Age squared/100	Wave 2		Wave 4		Married	Household size	arrennin arrennit	Employed		Other activities		1st wealth quartile		2nd wealth quartile		3rd wealth quartile		Î

Mental health
stage results:
nd second
First aı
Table 16:

		TGIDT	IG TO - COILUT	d mon nann	and enorvar	D			
Dependent variable:	EURO-D	family	formal	EURO-D	family	formal	EURO-D	family	formal
Parent in poor health		$0.194^{***}$	-0.002		$0.179^{***}$	0.003		$0.118^{***}$	0.002
(family care country)		(0.035)	(0.005)		(0.034)	(0.005)		(0.026)	(0.004)
Parent in poor health		0.000	$0.153^{***}$		0.000	$0.124^{***}$		$0.011^{*}$	$0.096^{***}$
(formal care country)		(0.008)	(0.035)		(0.007)	(0.029)		(0.007)	(0.021)
Observations	6,845	6,845	6,845	6,845	6,845	6,845	6,845	6,845	6,845
Within R-squared	-0.174	0.097	0.056	-0.193	0.083	0.035	-0.183	0.050	0.029
Unique individuals	3,009	3,009	3,009	3,009	3,009	3,009	3,009	3,009	3,009
Standard errors cluster	red at house	hold level in	parentheses	*** p<0.0	1, ** p < 0.0	15, * p < 0.1			
Pornections include and	o*conntry or	יפוווס משפ לר	100*con	ntw. interact	ion tarme				

continued from previous page Table 16 –

Regressions include age\*country and age squared/100\*country interaction terms. <sup>a</sup> Caregiver in family care country, <sup>b</sup> Caregiver in formal care country.

formal			-0.001	(0.003)	-0.003	(0.007)	0.007	(0.005)	0.025	(0.077)	-0.026	(0.065)	0.010	(0.034)	0.038	(0.081)	0.017	(0.025)	0.012	(0.007)	-0.004	(0.010)	-0.001	(0.009)	-0.004	(0.013)	-0.001	(0.011)	-0.006	(0.008)	-0.001	(0.001)	next page
family			0.011	(0.00)	-0.005	(0.013)	-0.008	(0.010)	-0.003	(0.014)	0.009	(0.007)	-0.001	(0.020)	-0.040	(0.072)	0.056	(0.037)	0.001	(0.010)	0.002	(0.017)	0.012	(0.013)	$0.063^{**}$	(0.026)	0.020	(0.021)	0.018	(0.018)	$0.114^{***}$	(0.026)	ntinued on
SAH	$\begin{array}{c} 0.410 \\ (0.522) \\ 0.712 \end{array}$	0.713 (0.705)	$0.163^{***}$	(0.020)	0.062	(0.046)	$0.107^{***}$	(0.037)	-0.215	(0.226)	0.169	(0.189)	$0.237^{**}$	(0.103)	$0.474^{*}$	(0.275)	-0.121	(0.122)	-0.011	(0.031)	0.009	(0.050)	-0.084**	(0.038)	-0.023	(0.073)	0.015	(0.055)	-0.005	(0.049)			ů
formal			0.002	(0.004)	0.014	(0.011)	-0.000	(0.007)	0.067	(0.106)	-0.081	(0.090)	0.041	(0.036)	0.098	(0.088)	-0.052	(0.043)	-0.004	(0.007)	$0.022^{*}$	(0.014)	-0.004	(0.012)	-0.009	(0.016)	-0.004	(0.014)	-0.002	(0.011)	0.002	(0.002)	
family			$0.017^{**}$	(0.008)	-0.007	(0.016)	0.007	(0.014)	0.008	(0.019)	$0.018^{*}$	(0.011)	-0.032	(0.026)	$-0.176^{**}$	(0.084)	$0.085^{**}$	(0.040)	-0.003	(0.013)	0.008	(0.024)	0.020	(0.018)	$0.096^{***}$	(0.028)	0.008	(0.023)	0.010	(0.021)	$0.184^{***}$	(0.034)	
HdS	$0.244 \\ (0.319) \\ 0.760 \\ 0.$	(0.509)	$0.161^{***}$	(0.020)	0.052	(0.046)	$0.108^{***}$	(0.036)	-0.235	(0.227)	0.192	(0.192)	$0.230^{**}$	(0.101)	$0.477^{*}$	(0.277)	-0.080	(0.124)	0.000	(0.030)	-0.007	(0.052)	-0.083**	(0.039)	-0.019	(0.071)	0.022	(0.053)	-0.004	(0.047)			
$formal^b$			-0.001	(0.005)	0.020	(0.013)	-0.010	(0.009)	0.073	(0.125)	-0.078	(0.103)	-0.007	(0.044)	-0.043	(0.108)	-0.034	(0.045)	-0.004	(0.008)	$0.031^{**}$	(0.015)	0.004	(0.015)	-0.006	(0.022)	0.013	(0.017)	0.008	(0.016)	$0.005^{**}$	(0.002)	
$family^a$			0.012	(0.009)	-0.019	(0.018)	0.020	(0.015)	-0.004	(0.021)	0.011	(0.011)	0.001	(0.030)	-0.050	(0.093)	0.069	(0.045)	0.017	(0.014)	-0.013	(0.025)	0.033	(0.020)	$0.071^{**}$	(0.032)	0.005	(0.024)	0.008	(0.023)	$0.198^{***}$	(0.034)	
HdS	$\begin{array}{c} 0.223 \\ (0.296) \\ 0.467 \end{array}$	(0.398)	$0.164^{***}$	(0.019)	0.053	(0.047)	$0.109^{***}$	(0.037)	-0.227	(0.225)	0.184	(0.188)	$0.246^{**}$	(0.104)	$0.514^{*}$	(0.279)	-0.088	(0.118)	-0.005	(0.031)	-0.003	(0.052)	-0.089**	(0.039)	-0.013	(0.067)	0.016	(0.053)	-0.008	(0.047)			
Dependent variable:	Caregiver (family care country)	Caregiver (formal care country)	Chronic conditions		ADL		IADL		Age		Age squared/ $100$		Wave 2		Wave 4		Married		Household size		Employed		Other activities		1st wealth quartile		2nd wealth quartile		3rd wealth quartile		Parent in poor health	(family care country)	

(HdS)
health
-perceived
Self
results:
stage
second
and
First
17:
Table

						0			
Dependent variable:	HdS	family	formal	HdS	family	formal	SAH	family	formal
Parent in poor health		0.002	$0.159^{***}$		0.002	$0.124^{***}$		0.002	$0.090^{***}$
(formal care country)		(0.003)	(0.034)		(0.003)	(0.027)		(0.002)	(0.019)
Observations	6,961	6,961	6,961	6,961	6,961	6,961	6,961	6,961	6,961
Within R-squared	0.035	0.098	0.055	0.035	0.085	0.035	0.034	0.050	0.028
Unique individuals	3,057	3,057	3,057	3,057	3,057	3,057	3,057	3,057	3,057
Standard errors cluster	ed at house	i level i	n parenthese	s. *** p<(	.01, ** p<	0.05, * p < 0.1			
- - -	•		+00 =/ -	•	•				

Table 17 – continued from previous page

Regressions include age\*country and age squared/100\*country interaction terms.  $^a$  Caregiver in family care country;  $^b$  Caregiver in formal care country.

formal	-0.001	(0.004) 0.000	(0.006)	(0.004)	0.039	-0.031 (U.Uoz)	(0.070)	-0.009	(0.034)	-0.008	(0.082)	0.019	(0.028)	0.013	(0.008)	-0.003	(0.010)	-0.003	(0.010)	-0.011	(0.013)	-0.006	(0.011)	-0.008	(0.008)	-0.001	(0.001)	next page
family	$0.013^{*}$	(0.008) -0.001	(0.016)	(0.013)	-0.005	(e10.0) 0.009	(0.008)	0.002	(0.022)	-0.028	(0.078)	0.062	(0.042)	0.003	(0.011)	0.001	(0.018)	0.012	(0.014)	$0.074^{***}$	(0.028)	0.031	(0.022)	0.024	(0.019)	$0.111^{***}$	(0.028)	ntinued on
GS 3.405 (2.927) -3.650	(2.465) -0.510***	(0.133) -0.256	(0.344)	(0.280)	1.710	(1.171)	(0.963)	-0.455	(0.538)	-2.078	(1.485)	0.021	(0.829)	0.112	(0.171)	0.026	(0.350)	-0.254	(0.250)	-0.630	(0.489)	-0.214	(0.348)	-0.314	(0.283)			Ŭ
formal	0.005	(0.005) 0.016	(0.012)	(0.007)	0.060	(0.112) -0.072	(0.095)	0.037	(0.038)	0.083	(0.095)	-0.057	(0.047)	-0.004	(0.008)	0.023	(0.014)	-0.006	(0.013)	-0.009	(0.018)	-0.004	(0.015)	-0.003	(0.012)	0.002	(0.002)	
family	$0.019^{**}$	(0.009) -0.000	(0.019)	(0.016)	0.007	(0.021)	(0.012)	-0.029	(0.028)	$-0.173^{*}$	(0.091)	$0.093^{**}$	(0.045)	-0.002	(0.015)	0.006	(0.025)	0.022	(0.019)	$0.100^{***}$	(0.030)	0.017	(0.025)	0.014	(0.022)	$0.183^{***}$	(0.036)	
GS 2.121 (1.742) -2.708	(1.822)-0.491***	(0.129) -0.214	(0.352)	(0.272)	1.697	(1.1.1)	(0.969)	-0.254	(0.542)	-1.553	(1.516)	-0.187	(0.812)	0.069	(0.161)	0.088	(0.348)	-0.265	(0.254)	-0.577	(0.458)	-0.132	(0.328)	-0.240	(0.273)			
formal <sup>b</sup>	0.002	(0.006) 0.023	(0.015)	(0.00)	0.095	(161.0) (161.0)	(0.109)	-0.030	(0.047)	-0.102	(0.116)	-0.032	(0.050)	-0.002	(0.008)	$0.030^{*}$	(0.016)	0.006	(0.016)	-0.007	(0.024)	0.015	(0.019)	0.008	(0.017)	$0.005^{**}$	(0.002)	
family <sup>a</sup>	0.014	(0.010) -0.017	(0.021)	(0.016)	-0.004	(0.013)	(0.012)	-0.001	(0.032)	-0.064	(0.100)	0.078	(0.050)	0.015	(0.015)	-0.019	(0.026)	0.031	(0.022)	$0.069^{**}$	(0.034)	0.008	(0.025)	0.008	(0.024)	$0.206^{***}$	(0.037)	
GS 1.912 (1.533) -2.246	(1.560) -0.485***	(0.127) -0.172	(0.354)	(0.267)	1.771	(1.107) -1.357	(0.971)	-0.479	(0.542)	-2.253	(1.493)	-0.058	(0.782)	0.040	(0.165)	0.143	(0.344)	-0.249	(0.256)	-0.486	(0.432)	-0.068	(0.326)	-0.199	(0.270)			
Dependent variable: Caregiver (family care country) Caregiver	(formal care country) Chronic conditions	ADL	TADL		Age	Age squared/100	•	Wave 2		Wave 4		Married		Household size		Employed		Other activities		1st wealth quartile		2nd wealth quartile		3rd wealth quartile		Parent in poor health	(family care country)	

(GS)
$\operatorname{strength}$
Grip
results:
stage
second
and
First
18:
Table

		Tal	DIE $1\delta - conti$	nued from [	orevious pa	lge Be			
Dependent variable:	GS	family	formal	GS	family	formal	GS	family	formal
Parent in poor health		0.003	$0.154^{***}$		0.002	$0.128^{***}$		0.002	$0.095^{***}$
(formal care country)		(0.003)	(0.036)		(0.003)	(0.029)		(0.002)	(0.020)
Observations	6,486	6,486	6,486	6,486	6,486	6,486	6,486	6,486	6,486
Within R-squared	0.036	0.096	0.055	0.037	0.082	0.036	0.014	0.050	0.031
Unique individuals	2,859	2,859	2,859	2,859	2,859	2,859	2,859	2,859	2,859
Standard errors clustere	ed at house	thold level in	a parentheses	).0>q *** .	11, ** p<0.	05, * p < 0.1			
			400						

Table 18 – continued from previous page

Regressions include age\* country and age squared/100\* country interaction terms.  $^a$  Caregiver in family care country,  $^b$  Caregiver in formal care country.

formal			-0.002	(0.003)	-0.003	(0.007)	0.007	(0.006)	0.025	(0.077)	-0.026	(0.065)	0.010	(0.034)	0.038	(0.081)	0.017	(0.025)	0.012	(0.008)	-0.004	(0.010)	-0.001	(0.009)	-0.003	(0.013)	-0.001	(0.011)	-0.006	(0.008)	-0.001	(0.001)	next page
family			0.010	(0.007)	-0.005	(0.013)	-0.008	(0.010)	-0.003	(0.014)	0.009	(0.007)	-0.001	(0.020)	-0.037	(0.072)	0.057	(0.037)	0.001	(0.010)	0.002	(0.017)	0.009	(0.013)	$0.065^{**}$	(0.026)	0.022	(0.021)	0.020	(0.019)	$0.116^{***}$	(0.026)	ntinued on
VF	$6.405^{*}$ (3.478) -1.999	(4.051)	$0.242^{*}$	(0.147)	-0.272	(0.281)	-0.302	(0.232)	1.479	(1.236)	-1.388	(0.997)	0.931	(0.773)	1.121	(2.084)	-0.678	(0.888)	0.014	(0.200)	-0.033	(0.369)	0.180	(0.320)	-0.652	(0.521)	-0.308	(0.374)	-0.213	(0.330)			Co
formal			0.002	(0.004)	0.014	(0.011)	-0.000	(0.007)	0.067	(0.106)	-0.081	(0.090)	0.041	(0.036)	0.098	(0.089)	-0.051	(0.043)	-0.004	(0.007)	$0.023^{*}$	(0.014)	-0.003	(0.012)	-0.009	(0.017)	-0.003	(0.014)	-0.001	(0.011)	0.002	(0.002)	
family			$0.017^{**}$	(0.008)	-0.007	(0.016)	0.007	(0.014)	0.006	(0.019)	$0.018^{*}$	(0.011)	-0.029	(0.026)	$-0.163^{*}$	(0.084)	$0.085^{**}$	(0.040)	-0.004	(0.013)	0.008	(0.024)	0.022	(0.018)	$0.097^{***}$	(0.028)	0.009	(0.023)	0.012	(0.021)	$0.180^{***}$	(0.034)	
VF	$4.145^{*}$ (2.149) -1.425	(2.992)	$0.245^{*}$	(0.145)	-0.251	(0.268)	$-0.398^{*}$	(0.226)	1.476	(1.271)	-1.470	(1.047)	1.086	(0.771)	1.627	(2.096)	-0.772	(0.885)	0.005	(0.193)	-0.011	(0.383)	0.147	(0.322)	-0.646	(0.505)	-0.208	(0.355)	-0.124	(0.321)			
$formal^b$			-0.001	(0.005)	0.020	(0.013)	-0.010	(0.009)	0.073	(0.125)	-0.079	(0.103)	-0.008	(0.045)	-0.045	(0.109)	-0.034	(0.045)	-0.004	(0.008)	$0.029^{*}$	(0.015)	0.004	(0.015)	-0.006	(0.022)	0.012	(0.017)	0.006	(0.016)	$0.005^{**}$	(0.002)	
$family^a$			0.012	(0.00)	-0.019	(0.018)	0.019	(0.015)	-0.004	(0.021)	0.010	(0.011)	0.002	(0.030)	-0.046	(0.094)	0.070	(0.045)	0.016	(0.014)	-0.013	(0.025)	0.031	(0.021)	$0.073^{**}$	(0.032)	0.006	(0.024)	0.010	(0.023)	$0.200^{***}$	(0.034)	
VF	$3.739^{**}$ (1.873) -1.132	(2.340)	$0.265^{*}$	(0.141)	-0.203	(0.273)	$-0.452^{**}$	(0.223)	1.505	(1.261)	-1.408	(1.028)	0.889	(0.765)	0.930	(2.055)	-0.648	(0.851)	-0.070	(0.186)	0.070	(0.386)	0.133	(0.317)	-0.512	(0.466)	-0.175	(0.350)	-0.105	(0.319)			
Dependent variable:	Caregiver (family care country) Caregiver	(formal care country)	Chronic conditions		ADL		IADL		Age		Age squared/100		Wave 2		Wave 4		Married		Household size		Employed		Other activities		1st wealth quartile		2nd wealth quartile		3rd wealth quartile		Parent in poor health	(family care country)	

(VF)
fluency
Verbal
results:
stage
second
and
First
19:
Table

	formal	$0.090^{***}$	(0.020)	6,935	0.028	3,044		
	family	0.002	(0.002)	6,935	0.051	3,044		
	VF			6,935	-0.019	3,044	0.1	
202	formal	$0.124^{***}$	(0.027)	6,935	0.034	3,044	< 0.05, * p <	
d approad	family	0.002	(0.003)	6,935	0.083	3,044	< 0.01, ** p	
11011 000	VF			6,935	0.004	3,044	es. *** p.	
	formal	$0.156^{***}$	(0.034)	6,935	0.053	3,044	in parenthes	+0011
	family	0.002	(0.003)	6,935	0.099	3,044	ehold level	
	VF			6,935	0.019	3,044	ed at hous	
	Dependent variable:	Parent in poor health	(formal care country)	Observations	Within R-squared	Unique individuals	Standard errors clustere	-

Table 19 – continued from previous page

Regressions include age\*country and age squared/100\*country interaction terms.  $^a$  Caregiver in family care country;  $^b$  Caregiver in formal care country.

formal				-0.001	(0.003)	-0.003	(0.007)	0.007	(0.006)	0.025	(0.077)	-0.026	(0.065)	0.010	(0.034)	0.038	(0.081)	0.017	(0.025)	0.012	(0.008)	-0.004	(0.010)	-0.002	(0.009)	-0.004	(0.013)	-0.001	(0.011)	-0.006	(0.008)	-0.001	(0.001)
family				0.010	(0.007)	-0.005	(0.013)	-0.008	(0.010)	-0.004	(0.014)	0.009	(0.007)	-0.000	(0.020)	-0.036	(0.072)	0.057	(0.037)	0.001	(0.010)	0.002	(0.017)	0.009	(0.013)	$0.064^{**}$	(0.026)	0.022	(0.021)	0.020	(0.018)	$0.115^{**}$	(0.026)
STWR	-0.583 (0.985)	0.797	(1.140)	0.013	(0.041)	0.015	(0.092)	0.014	(0.069)	0.333	(0.339)	-0.198	(0.279)	0.013	(0.186)	-0.190	(0.503)	$0.651^{**}$	(0.279)	-0.067	(0.054)	0.076	(0.097)	-0.033	(0.086)	0.090	(0.153)	0.114	(0.123)	-0.100	(0.095)		
formal				0.002	(0.004)	0.014	(0.011)	-0.000	(0.007)	0.067	(0.106)	-0.081	(0.090)	0.041	(0.036)	0.097	(0.088)	-0.051	(0.043)	-0.004	(0.007)	$0.022^{*}$	(0.014)	-0.004	(0.012)	-0.009	(0.016)	-0.003	(0.014)	-0.002	(0.011)	0.002	(0.002)
family				$0.017^{**}$	(0.008)	-0.007	(0.016)	0.007	(0.014)	0.006	(0.019)	$0.018^{*}$	(0.011)	-0.028	(0.026)	$-0.162^{*}$	(0.084)	$0.085^{**}$	(0.040)	-0.004	(0.013)	0.008	(0.024)	0.022	(0.018)	$0.096^{***}$	(0.028)	0.009	(0.023)	0.012	(0.021)	$0.179^{***}$	(0.033)
STWR	-0.386	0.579	(0.817)	0.011	(0.042)	0.005	(0.092)	0.027	(0.069)	0.318	(0.343)	-0.170	(0.287)	-0.013	(0.184)	-0.258	(0.503)	$0.694^{**}$	(0.284)	-0.057	(0.053)	0.061	(0.100)	-0.029	(0.088)	0.092	(0.152)	0.106	(0.121)	-0.111	(0.092)		
$formal^b$				-0.001	(0.005)	0.020	(0.013)	-0.010	(0.00)	0.073	(0.125)	-0.079	(0.103)	-0.007	(0.044)	-0.043	(0.108)	-0.034	(0.045)	-0.004	(0.008)	$0.031^{**}$	(0.015)	0.004	(0.015)	-0.006	(0.022)	0.013	(0.017)	0.008	(0.016)	$0.005^{**}$	(0.002)
$family^a$				0.012	(0.009)	-0.019	(0.018)	0.019	(0.015)	-0.004	(0.021)	0.011	(0.011)	0.003	(0.030)	-0.045	(0.093)	0.070	(0.045)	0.016	(0.014)	-0.013	(0.025)	0.031	(0.021)	$0.073^{**}$	(0.032)	0.006	(0.024)	0.010	(0.023)	$0.200^{***}$	(0.034)
STWR	-0.352 (0.563)	0.452	(0.658)	0.011	(0.042)	-0.001	(0.092)	0.036	(0.071)	0.320	(0.340)	-0.184	(0.281)	0.025	(0.187)	-0.136	(0.509)	$0.672^{**}$	(0.282)	-0.051	(0.055)	0.053	(0.101)	-0.030	(0.089)	0.078	(0.145)	0.097	(0.121)	-0.116	(0.093)		
Dependent variable:	Caregiver (family care country)	Caregiver	(formal care country)	Chronic conditions		ADL		IADL		Age		Age squared/100		Wave 2		Wave 4		Married		Household size		Employed		Other activities		1st wealth quartile		2nd wealth quartile		3rd wealth quartile		Parent in poor health	(family care country)

(STWR)
recall
word
Short-term
results:
stage
second
and
$\operatorname{First}$
20:
Table

		TAUJ	nno – oz a	manii nanii	f snorvard	Jage			
Dependent variable:	STWR	family	formal	STWR	family	formal	STWR	family	formal
Parent in poor health		0.002	$0.159^{***}$		0.002	$0.124^{***}$		0.002	$0.090^{***}$
(formal care country)		(0.003)	(0.034)		(0.003)	(0.027)		(0.002)	(0.019)
Observations	6,950	6,950	6,950	6,950	6,950	6,950	6,950	6,950	6,950
Within R-squared	0.037	0.098	0.055	0.041	0.083	0.035	0.042	0.050	0.028
Unique individuals	3,051	3,051	3,051	3,051	3,051	3,051	3,051	3,051	3,051
Standard errors cluster	red at hous	ehold level	in parenthe	ses. *** p.	<0.01, ** I	o<0.05, * p<	< 0.1		
Regressions include ag	e*country a	und age squ	ared/100*c	ountry inte	eraction te	rms.			
<sup><math>a</math></sup> Caregiver in family c	care country	r; <sup>b</sup> Caregi	ver in forma	l care cour	itry.				

Table 20 – continued from previous page

TWR family formal ).545 .055) 1.502 224)	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{ccccccc} .052 & -0.005 & -0.003 \\ .098) & (0.013) & (0.007) \end{array}$	$\begin{array}{ccccccc}052 & -0.005 & -0.003 \\098) & (0.013) & (0.007) \\046 & -0.008 & 0.007 \\090) & (0.010) & (0.006) \\113 & -0.004 & 0.025 \\422) & (0.014) & (0.077) \\ \end{array}$	$\begin{array}{ccccccc}$	$\begin{array}{cccccccc}$	$\begin{array}{cccccccc}$	$\begin{array}{cccccccc}$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{ccc} 0.002 & -0.0 \\ (0.004) & (0.0 \end{array}$	$\begin{array}{c} 0.014 & 0.0 \\ (0.011) & (0.02) \end{array}$	$\begin{array}{ccc} -0.000 & -0.0 \\ (0.007) & (0.07) & (0.07) \\ 0.067 & 0.1 \\ 0.1060 & (0.44) \end{array}$	-0.081 0.1. (0.090) (0.3 0.041 -0.2	$\begin{array}{c} 0.036 \\ 0.036 \\ 0.097 \\ 0.06 \\ 0.06 \\ 0.06 \end{array}$	$\begin{array}{c} -0.051 \\ 0.043 \\ -0.004 \\ -0.004 \\ -0.0 \end{array}$	$\begin{array}{c} (0.007) & (0.0\\ 0.022^{*} & -0.1\\ (0.014) & (0.12\\ 0.014 & 0.02\\ 0.004 & 0.004\\ 0.004 & 0$	-0.004 0.0 (0.012) (0.0 -0.009 -0.0 (0.016) (0.10	-0.003 $-0.0(0.014) (0.1.$	-0.002 -0.27 (0.011) (0.1 0.002 (0.02) (0.2)	
	0.017** (0.008) -0.007 (0.016)	$\begin{array}{c} 0.007\\ (0.014)\\ 0.006\\ (0.019)\end{array}$	$0.018^{*}$ (0.011) -0.028	(0.026) -0.162* (0.084)	$0.085^{**}$ (0.040) -0.004	(0.013) 0.008 (0.024)	$\begin{array}{c} 0.022\\ (0.018)\\ 0.096^{***}\\ (0.028) \end{array}$	(0.023)	$\begin{array}{c} 0.012\\ (0.021)\\ 0.179^{***}\\ (0.033)\end{array}$	
-0.330 (0.678) -1.097 (0.904)	$\begin{array}{c} -0.031\\ (0.047)\\ 0.072\\ (0.096) \end{array}$	-0.051 (0.089) 0.153 (0.443)	0.084 (0.371) -0.186	(0.221) -0.671 (0.604)	0.488 (0.456) -0.060	(0.067) -0.133 (0.123)	$\begin{array}{c} 0.059\\ (0.094)\\ -0.051\\ (0.163)\end{array}$	(0.129)	(0.110)	
	$\begin{array}{c} -0.001 \\ (0.005) \\ 0.020 \\ (0.013) \end{array}$	-0.010 (0.009) 0.073 (0.125)	-0.079 (0.103) -0.007	(0.044) -0.043 (0.108)	-0.034 (0.045) -0.004	(0.008) $0.031^{**}$ (0.015)	$\begin{array}{c} 0.004\\ (0.015)\\ -0.006\\ (0.022) \end{array}$	(0.013)	$\begin{array}{c} 0.008\\ (0.016)\\ 0.005^{**}\\ (0.002) \end{array}$	
	$\begin{array}{c} 0.012 \\ (0.009) \\ -0.019 \\ (0.018) \end{array}$	$\begin{array}{c} 0.019 \\ (0.015) \\ -0.004 \\ (0.021) \end{array}$	(0.011) (0.011) 0.003	(0.030) -0.045 (0.093)	(0.070) (0.045) 0.016	(0.014) -0.013 (0.025)	$\begin{array}{c} 0.031\\ (0.021)\\ 0.073^{**}\\ (0.032) \end{array}$	(0.024)	$\begin{array}{c} 0.010\\ (0.023)\\ 0.200^{***}\\ (0.034) \end{array}$	
-0.286 (0.608) -0.856 (0.688)	-0.036 (0.046) 0.071 (0.099)	-0.055 (0.090) 0.139 (0.434)	0.102 (0.361) -0.226	(0.218) -0.775 (0.596)	0.507 (0.456) -0.053	(0.070) -0.138 (0.123)	$\begin{array}{c} 0.099\\ (0.094)\\ -0.057\\ (0.156) \end{array}$	-0.016 (0.129)	$-0.263^{+1}$ (0.109)	
Dependent variance. Caregiver Caregiver formal care country)	Chronic conditions	IADL Age	Age squared/100 Wave 2	Wave 4	Married Household size	Employed	Other activities 1st wealth quartile	2nd wealth quartile	ord wealth quartile Parent in poor health (family care country)	

Table 21: First and second stage results: Long-term word recall (LTWR)

		Tau	10 71 - COII01	innen monti	nd enorvard	age of the second			
Dependent variable:	LTWR	family	formal	LTWR	family	formal	LTWR	family	formal
Parent in poor health		0.002	$0.159^{***}$		0.002	$0.124^{***}$		0.002	$0.090^{***}$
(formal care country)		(0.003)	(0.034)		(0.003)	(0.027)		(0.002)	(0.019)
Observations	6,950	6,950	6,950	6,950	6,950	6,950	6,950	6,950	6,950
Within R-squared	0.061	0.098	0.055	0.056	0.083	0.035	0.054	0.050	0.028
Unique individuals	3,051	3,051	3,051	3,051	3,051	3,051	3,051	3,051	3,051
Standard errors cluster	ed at house	shold level	in parenthes	tes. *** p<	$0.01, ^{**} p <$	(0.05, * p < 0	.1		
- - - -	•	-	+00 =/ -	•					

Table 21 – continued from previous page

Regressions include age\*country and age squared/100\*country interaction terms.  $^a$  Caregiver in family care country;  $^b$  Caregiver in formal care country.

$ \begin{array}{cccccc} & 0.411 & 0.457 & 0.457 & 0.708 & 0.708 \\                                   $	endent variable:	numeracy	$family^a$	$formal^b$	numeracy	family	formal	numeracy	family	formal
$ \begin{array}{ccccc} \mbox{matrix} & (0.220) & (0.240) & (0.240) & (0.16) & (0.001) & (0.16) & (0.001) & (0.002) & (0.160) & (0.001) & (0.002) & $	r	-0.411			-0.457			-0.708		
$ \begin{array}{cccccc} \mbox{acc} & (0.270) & (0.333) & (0.456) & (0.011) & (0.005) & (0.015) & (0.003) & (0.011) & (0.003) & (0.011) & (0.003) & (0.001) & (0.003) & $	are country) r	(0.262) 0.060			(0.290) 0.073			(0.450) 0.106		
$ \begin{array}{c} {\rm conditions} & -0.021 & 0.013 & -0.001 & -0.018 & 0.017 & 0.002 & -0.018 & 0.011 & -0.002 \\ -0.017 & 0.009 & 0.005 & 0.0038 & 0.011 & 0.007 & 0.003 & 0.003 \\ -0.017 & 0.020 & 0.020 & -0.013 & 0.014 & 0.015 & 0.007 & 0.003 \\ -0.017 & 0.020 & 0.020 & -0.011 & 0.007 & 0.001 & 0.011 & 0.003 & 0.008 \\ -0.011 & 0.001 & 0.013 & 0.013 & 0.014 & 0.007 & 0.003 & 0.008 \\ -0.011 & 0.001 & 0.010 & 0.011 & 0.007 & 0.003 & 0.001 \\ -0.022 & 0.004 & 0.011 & 0.003 & 0.011 & 0.007 & 0.003 & 0.003 \\ -0.0131 & 0.001 & 0.011 & 0.013 & 0.013 & 0.0133 & 0.024 \\ -0.0131 & 0.001 & 0.011 & 0.013 & 0.0133 & 0.0133 & 0.024 \\ -0.131 & 0.001 & 0.011 & 0.013 & 0.0133 & 0.0133 & 0.007 & 0.007 \\ 0.131 & 0.001 & 0.013 & 0.013 & 0.0133 & 0.0133 & 0.001 & 0.011 \\ 0.100 & 0.030 & 0.013 & 0.013 & 0.0133 & 0.0133 & 0.004 & 0.011 & 0.001 \\ 0.100 & 0.030 & 0.013 & 0.013 & 0.0133 & 0.013 & 0.024 & 0.011 & 0.001 \\ 0.100 & 0.030 & 0.0043 & 0.014 & 0.014 & 0.013 & 0.025 \\ 0.013 & 0.004 & 0.011 & 0.003 & 0.024 & 0.011 & 0.003 & 0.025 \\ 0.013 & 0.004 & 0.013 & 0.003 & 0.004 & 0.011 & 0.001 & 0.012 \\ 0.100 & 0.030 & 0.003 & 0.004 & 0.013 & 0.007 & 0.001 & 0.001 \\ 0.014 & 0.011 & 0.003 & 0.012 & 0.003 & 0.002 & 0.001 & 0.001 \\ 0.014 & 0.014 & 0.014 & 0.013 & 0.002 & 0.001 & 0.002 \\ 0.003 & 0.004 & 0.001 & 0.002 & 0.001 & 0.002 \\ 0.003 & 0.004 & 0.001 & 0.002 & 0.001 & 0.001 \\ 0.014 & 0.003 & 0.013 & 0.003 & 0.002 & 0.001 & 0.002 \\ 0.013 & 0.003 & 0.001 & 0.003 & 0.002 & 0.001 & 0.002 \\ 0.004 & 0.004 & 0.001 & 0.002 & 0.001 & 0.001 \\ 0.014 & 0.003 & 0.002 & 0.003 & 0.002 & 0.001 & 0.002 \\ 0.014 & 0.003 & 0.002 & 0.003 & 0.002 & 0.001 & 0.001 & 0.001 \\ 0.014 & 0.003 & 0.001 & 0.002 & 0.001 & 0.002 & 0.001 \\ 0.014 & 0.003 & 0.001 & 0.002 & 0.001 & 0.002 & 0.001 \\ 0.014 & 0.003 & 0.002 & 0.003 & 0.002 & 0.001 & 0.002 & 0.001 \\ 0.014 & 0.003 & 0.002 & 0.003 & 0.002 & 0.001 & 0.002 & 0.001 \\ 0.014 & 0.003 & 0.002 & 0.003 & 0.002 & 0.003 & 0.001 & 0.002 & 0.001 \\ 0.014 & 0.003 & 0.002 & 0.003 & 0.002 & 0.003 & 0.002 & 0.001 & 0.002 & 0.001 \\ 0.014 & 0.003 &$	are country)	(0.270)			(0.333)			(0.456)		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	conditions	-0.021	0.013	-0.001	-0.018	$0.017^{**}$	0.002	-0.018	0.011	-0.002
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.015)	(0.00)	(0.005)	(0.015)	(0.008)	(0.004)	(0.015)	(0.007)	(0.003)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		-0.017	-0.020	0.020	-0.012	-0.007	0.014	-0.012	-0.005	-0.003
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.037)	(0.018)	(0.013)	(0.038)	(0.016)	(0.011)	(0.039)	(0.013)	(0.007)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		-0.006	0.020	-0.009	-0.011	0.007	-0.000	-0.021	-0.008	0.008
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.041)	(0.015)	(000.0)	(0.042)	(0.014)	(0.007)	(0.044)	(0.010)	(0.005)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		0.239	-0.004	0.076	0.243	0.007	0.065	0.240	-0.003	0.024
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.167)	(0.022)	(0.125)	(0.167)	(0.019)	(0.106)	(0.166)	(0.014)	(0.077)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ared/100	-0.217	0.011	-0.080	-0.212	$0.018^{*}$	-0.080	-0.217	0.009	-0.025
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.134)	(0.011)	(0.103)	(0.135)	(0.011)	(0.090)	(0.133)	(0.007)	(0.065)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		0.131	0.001	-0.012	0.113	-0.030	0.043	0.128	-0.001	0.011
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.100)	(0.030)	(0.044)	(0.099)	(0.026)	(0.036)	(0.100)	(0.020)	(0.034)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		0.213	-0.050	-0.054	0.147	$-0.166^{**}$	0.103	0.198	-0.040	0.041
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.260)	(0.094)	(0.108)	(0.261)	(0.084)	(0.089)	(0.259)	(0.072)	(0.082)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		0.162	0.069	-0.034	0.173	$0.084^{**}$	-0.051	0.169	0.056	0.017
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		(0.120)	(0.045)	(0.045)	(0.123)	(0.040)	(0.043)	(0.120)	(0.037)	(0.025)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ld size	-0.004	0.017	-0.003	-0.012	-0.003	-0.004	-0.011	0.001	0.012
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.025)	(0.014)	(0.008)	(0.024)	(0.013)	(0.007)	(0.025)	(0.010)	(0.007)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	þ	0.042	-0.013	$0.031^{**}$	0.051	0.008	$0.022^{*}$	0.051	0.002	-0.004
tivities $0.007$ $0.033$ $0.006$ $0.005$ $0.024$ $-0.004$ $0.002$ $0.011$ $-0.002$ th quartile $0.0341$ $(0.021)$ $(0.015)$ $(0.0134)$ $(0.012)$ $(0.013)$ $(0.013)$ th quartile $0.036$ $0.071^{**}$ $-0.006$ $0.050$ $0.095^{***}$ $-0.009$ $0.051$ $(0.013)$ $(0.003)$ th quartile $0.003$ $0.071^{**}$ $-0.006$ $0.067$ $(0.023)$ $(0.016)$ $(0.067)$ $(0.023)$ $(0.013)$ th quartile $0.003$ $0.005$ $0.017$ $(0.020)$ $(0.023)$ $(0.014)$ $(0.057)$ $(0.021)$ $(0.013)$ th quartile $-0.044$ $0.002$ $0.017$ $(0.017)$ $(0.023)$ $(0.014)$ $(0.050)$ $(0.021)$ $(0.011)$ th quartile $-0.004$ $0.007$ $-0.002$ $0.010$ $-0.001$ $0.007$ $0.018$ $-0.001$ th quartile $0.033$ $(0.014)$ $(0.023)$ $(0.014)$ $(0.021)$ $(0.011)$ th quartile $0.003$ $0.07$ $-0.002$ $0.010$ $-0.001$ $0.007$ $0.018$ $-0.006$ n poor health $0.200^{***}$ $0.065^{***}$ $0.002$ $0.010$ $(0.011)$ $(0.044)$ $(0.019)$ $(0.003)$ un poor health $0.200^{***}$ $0.005^{***}$ $0.002$ $0.010$ $(0.011)$ $(0.044)$ $(0.019)$ $(0.003)$ h poor health $0.0034$ $(0.023)$ $(0.016)$ $(0.021)$ $(0.011)$ $(0.044)$ $(0.019)$ $(0.003)$		(0.051)	(0.025)	(0.015)	(0.051)	(0.024)	(0.014)	(0.049)	(0.017)	(0.010)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ctivities	0.007	0.033	0.006	0.005	0.024	-0.004	0.002	0.011	-0.002
th quartile $0.036$ $0.071^{**}$ $-0.006$ $0.050$ $0.095^{***}$ $-0.009$ $0.051$ $0.063^{***}$ $-0.004$ (0.064) $(0.032)$ $(0.022)$ $(0.068)$ $(0.028)$ $(0.016)$ $(0.067)$ $(0.026)$ $(0.013)$ (14) quartile $0.003$ $0.005$ $0.011$ $0.006$ $0.007$ $-0.003$ $0.017$ $0.021$ $-0.001$ (16) $(0.048)$ $(0.024)$ $(0.017)$ $(0.049)$ $(0.023)$ $(0.014)$ $(0.050)$ $(0.021)$ $(0.011)$ (17) $(0.011)$ $(0.011)$ $(0.043)$ $(0.023)$ $(0.014)$ $(0.021)$ $(0.011)$ (10) $(0.011)$ $(0.043)$ $(0.023)$ $(0.014)$ $(0.021)$ $(0.011)$ $(0.011)$ (10) $(0.043)$ $(0.023)$ $(0.016)$ $(0.044)$ $(0.021)$ $(0.019)$ $(0.08)$ (10) $(0.023)$ $(0.016)$ $(0.044)$ $(0.021)$ $(0.011)$ $(0.044)$ $(0.019)$ $(0.008)$ (10) $(0.008)$ $(0.005^{**})$ $(0.016)$ $(0.034)$ $(0.034)$ $(0.022)$ $(0.014)$ $(0.019)$ $(0.008)$ (10) $(0.034)$ $(0.022)$ $(0.034)$ $(0.02)$ $(0.034)$ $(0.02)$ $(0.020)$ $(0.026)$ $(0.011)$ (16) $(0.001)$		(0.034)	(0.021)	(0.015)	(0.034)	(0.018)	(0.012)	(0.033)	(0.013)	(0.00)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	th quartile	0.036	$0.071^{**}$	-0.006	0.050	$0.095^{***}$	-0.009	0.051	$0.063^{**}$	-0.004
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		(0.064)	(0.032)	(0.022)	(0.068)	(0.028)	(0.016)	(0.067)	(0.026)	(0.013)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	lth quartile	0.003	0.005	0.011	0.006	0.007	-0.003	0.017	0.021	-0.001
th quartile $-0.004$ $0.008$ $0.007$ $-0.002$ $0.010$ $-0.001$ $0.007$ $0.018$ $-0.006$ (0.043) $(0.023)$ $(0.016)$ $(0.044)$ $(0.011)$ $(0.014)$ $(0.019)$ $(0.008)n poor health 0.200^{***} 0.005^{***} 0.005^{***} 0.180^{***} 0.002 0.116^{***} -0.001care country) (0.034) (0.02) (0.002) (0.020) (0.026) (0.001)$		(0.048)	(0.024)	(0.017)	(0.049)	(0.023)	(0.014)	(0.050)	(0.021)	(0.011)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	th quartile	-0.004	0.008	0.007	-0.002	0.010	-0.001	0.007	0.018	-0.006
n poor health $0.200^{***}$ $0.005^{**}$ $0.002^{***}$ $0.002$ $0.116^{***}$ $-0.001$ earce country) $(0.034)$ $(0.02)$ $(0.024)$ $(0.002)$ $(0.026)$ $(0.021)$		(0.043)	(0.023)	(0.016)	(0.044)	(0.021)	(0.011)	(0.044)	(0.019)	(0.008)
(0.034)  (0.034)  (0.002)  (0.034)  (0.002)  (0.026)  (0.021)	a poor health		$0.200^{***}$	$0.005^{**}$		$0.180^{***}$	0.002		$0.116^{***}$	-0.001
	care country)		(0.034)	(0.002)		(0.034)	(0.002)		(0.026)	(0.001)

		Ϋ́́,		nd more popul	and morro	>			
Dependent variable:	numeracy	family	formal	numeracy	family	formal	numeracy	family	formal
Parent in poor health		0.002	$0.155^{***}$		0.001	$0.125^{***}$		0.002	$0.091^{***}$
(formal care country)		(0.003)	(0.034)		(0.003)	(0.028)		(0.002)	(0.020)
Observations	6,948	6,948	6,948	6,948	6,948	6,948	6,948	6,948	6,948
Within R-squared	-0.026	0.099	0.054	-0.033	0.083	0.035	-0.042	0.050	0.028
Unique individuals	3,051	3,051	3,051	3,051	3,051	3,051	3,051	3,051	3,051
Standard errors cluster	ed at househ	old level in	parentheses	. *** p<0.01	, ** p<0.05	5, * p<0.1			

Table 22 – continued from previous page

Regressions include age\*country and age squared/100\*country interaction terms.  $^a$  Caregiver in family care country;  $^b$  Caregiver in formal care country.

	All co	untries	Formal can	re countrie
Dependent variable:	EURO-D	caregiver	EURO-D	caregiver
Caregiver	0.950		$1.702^{*}$	
	(0.993)		(0.887)	
Parent in poor health	0.203	$0.183^{***}$	0.109	$0.150^{***}$
	(0.217)	(0.026)	(0.209)	(0.035)
Parental loss	$0.233^{**}$	$-0.076^{***}$	0.122	-0.134***
	(0.117)	(0.025)	(0.174)	(0.036)
Chronic conditions	$0.228^{***}$	0.007	$0.223^{***}$	-0.011
	(0.059)	(0.010)	(0.076)	(0.013)
ADL	-0.058	0.005	-0.098	0.048
	(0.145)	(0.023)	(0.187)	(0.033)
IADL	$0.294^{*}$	0.017	0.166	-0.015
	(0.158)	(0.018)	(0.129)	(0.028)
Age	-0.231	-0.005	-0.304	-0.010
	(0.475)	(0.126)	(0.513)	(0.128)
Age squared/100	0.132	-0.005	0.135	0.010
	(0.380)	(0.103)	(0.405)	(0.104)
Wave 2	-0.046	0.001	0.062	-0.021
	(0.271)	(0.053)	(0.310)	(0.056)
Wave 4	0.261	-0.090	0.776	-0.159
	(0.745)	(0.142)	(0.863)	(0.155)
Married	-0.557	0.020	-0.322	-0.065
	(0.403)	(0.069)	(0.381)	(0.085)
Household size	0.038	0.016	0.134	-0.009
	(0.096)	(0.016)	(0.122)	(0.024)
Employed	0.097	0.017	$-0.342^{*}$	$0.071^{*}$
	(0.145)	(0.030)	(0.198)	(0.036)
Other activities	-0.117	0.041	0.062	0.026
	(0.113)	(0.026)	(0.136)	(0.032)
1st wealth quartile	0.052	$0.065^{*}$	-0.357	-0.007
	(0.196)	(0.038)	(0.223)	(0.053)
2nd wealth quartile	-0.122	0.021	-0.206	0.038
	(0.156)	(0.029)	(0.194)	(0.040)
3rd wealth quartile	-0.175	0.016	-0.200	0.013
	(0.131)	(0.028)	(0.188)	(0.037)
Single parent		$0.213^{***}$		0.316***
		(0.052)		(0.072)
Observations	6,845	6,845	4,100	4,100
Within R-squared	0.005	0.088	-0.123	0.084
Unique individuals	3,009	3,009	1,755	1,755
Exogeneity (p-value)	0.358		0.031	
First stage F statistic	17.06		19.22	

Table 23: FE-IV results for mental health using "single parent" as alternative instrument

Standard errors clustered at household level in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1. Regressions include age\*country and age squared/100\*country interaction terms.